

## Consumption Feedback and Water Saving: A Field Intervention Evaluation in the Metropolitan Area of Milan

Stefano Clò<sup>1</sup> · Tommaso Reggiani<sup>2</sup> · Sabrina Ruberto<sup>3</sup>

Accepted: 9 May 2024 © The Author(s) 2024

## Abstract

This paper investigates whether informative feedback on consumption can nudge water saving. We launched a five-month online information campaign which involved around 1,000 households located in the province of Milan (Italy) with a smart meter. A group of households received monthly reports via email on their per capita daily average water consumption, including a social comparison component. The Intention to Treat (ITT) analysis shows that, compared to a benchmark group, the units exposed to the intervention reduced their per capita water consumption by around 6% (25.8 liters per day or 6.8 gallons). Being able to observe the email opening rate, we find that the ITT effect is mainly driven by complying units. Through an Instrumental Variable approach, we estimated a Local Average Treatment Effect equal to 54.9 liters per day of water saving. A further Regression Discontinuity Design analysis shows that different feedback on consumption class size differentially affected water saving at the margin. We also found that the additional water saving increased with the number of monthly reports, though it did not persist two months after the campaign expired.

Keywords Water saving  $\cdot$  Nudging  $\cdot$  Field intervention  $\cdot$  Online information campaign  $\cdot$  ITT and LATE

JEL Classification  $C93 \cdot H41 \cdot L95 \cdot Q25$ 

Stefano Clò stefano.clo@unifi.it

> Tommaso Reggiani reggianit@cardiff.ac.uk

Sabrina Ruberto sabrina.ruberto@unicz.it

<sup>1</sup> Department of Economics and Management, University of Florence, Florence, Italy

<sup>2</sup> Cardiff University | IZA | Masaryk University, Cardiff Business School, Cardiff, UK

<sup>&</sup>lt;sup>3</sup> Department of Law, Economics and Sociology, University Magna Graecia of Catanzaro, Catanzaro, Italy

## 1 Introduction

Water scarcity is already affecting a quarter of the world's population, causing economic damage (Franzke 2021), and negative consequences on human health and well-being (Ebi and Bowen 2016). Among the different actions aimed at mitigating this issue, the reduction of excessive water consumption, which contributes to local water stress, has become a primary sustainability objective.<sup>1</sup> With respect to this, behavioral nudges have been increasingly acknowledged as powerful and cost-effective actions that can supplement or replace traditional economic levers<sup>2</sup> to correct market failures, engage people in pro-social behavior, and align them with socio-valuable goals (World Bank 2015; Benartzi et al. 2017). Academic literature has offered wide and robust evidence on the behavioral sciences' capacity to correct cognitive biases while preserving fundamental individual freedom of choice (Sunstein 2018; Thaler & Sunstein 2008).

Behavioral insights are particularly relevant to the water sector. In various countries, water tariffs are regulated, thus limiting the possibility of water companies to leverage water saving by increasing the related price. Moreover, being a sector of general interest, a rise in price might inhibit goals of accessibility and service universalization. Non-monetary measures are thus likely to be politically more effective in promoting water conservation attitudes than traditional economic measures. On top of that, in various countries, people do not have access to updated and reliable information on their daily water consumption. This can partly be traced back to the widespread technological backwardness in the data collection and communication systems. Data on domestic water consumption can be observed directly via water meters, or indirectly via the water bill. Both ways imply non-negligible searching and evaluating costs. Concerning the former, in many countries (Italy included), homes are usually equipped with analogue water meters, which can be installed outside private homes (or, internally, but not in easily visible places). Moreover, water meters report households' cumulative consumption, limiting users' knowledge of their daily consumption. Concerning the latter, water bills usually report the total amount of households' aggregated water consumption over a given period (e.g. a quarter). This information might not be easy to understand, evaluate and compare. Moreover, since selfreported communications or door-to-door readings occur sporadically, water billing is usually calculated according to estimated rather than real data.<sup>3</sup>

Within this framework, the main goal of this research is to assess the extent to which informative feedback on water consumption can nudge water savings.

**The Information Campaign** To address our research question, we designed an online information campaign which involved around 1,000 households in the metropolitan area of Milan, over a five-month period, beginning in September 2021 and lasting until January 2022. The campaign was launched in partnership with the CAP Holding water company,

<sup>&</sup>lt;sup>1</sup> Among the United Nations' Sustainable Development Goals (SDGs), Goal 12 – 'Responsible Consumption and Production' – calls for effective and timely actions aimed at promoting sustainable behavior.

<sup>&</sup>lt;sup>2</sup> In this field, traditional economic levers include increasing water prices, command and control instruments, and market-based incentives aimed at inducing the adoption of water saving technologies (aerated jet breakers, double button discharge, electronic faucets with sensors and photocells) and more efficient domestic appliances.

<sup>&</sup>lt;sup>3</sup> Condominiums are often equipped with a single meter and the bills are divided according to the size of the apartments, and not according to the consumption of the individual condominiums, which are not registered.

one of the main Italian water companies, which manages the integrated water services in the metropolitan area of Milan and in other provinces of the Lombardy region. The CAP Holding water company was the first Italian company to start the replace process of the analogue water meters with electronic smart meters. This technology allowed for an automated, remote collection of the households' water consumption data, which was elaborated for the information campaign. In particular, treated units received via email on a monthly basis their *water consumption diary:* a brief report on the households' water consumption, including a social comparison component. Households were ranked according to their water consumption on a 1–1000 scale and were informed of both their ranking position and the related consumption category: households belonging to the first, second and third consumption tertiles, were ranked 'low', 'medium' and 'high' accordingly. Details on the information campaign are reported in Sect. 3.

**The Empirical Strategy** We focused our analysis on a sample of more than 13,800 households distributed over 45 municipalities in the metropolitan area of Milan (Italy) that, at the time of the implementation of the intervention, were already equipped with smart meters remotely collecting data on their water consumption. The sample is not determined by selfselection, since families could not apply for smart meters and the water company was not replacing the conventional meters following any pre-determined geographical criteria.

Within this sample, we were legally bound to send informative material only to around 1,000 households, which, at the time of signing their water contract, gave their consent to receive informative and ancillary communication from the water company.

The CAP Holding water company was committed to involve as many households as possible, therefore it firmly required to send the information campaign to all the 1,000 eligible households. Due to this exogenous constraint, a RCT could not represent a viable option, and we had to adopt an alternative empirical strategy. We initially considered as the treated group the entire subset of eligible households (those equipped with a smart meter and that had provided privacy consent), while the remaining households (those equipped with a smart meter, that had not provided privacy consent) were included into the control group.

Selection into the informative condition was based on a general willingness to receive ancillary communication, with this consent being given several years before the information campaign. For this reason, the adopted selection criterion is likely to be orthogonal to the intervention itself. Nevertheless, we are aware that people voluntarily decided whether or not to give their consent to be advertised. Therefore, following previous relevant literature, aiming to assess the impact of the information campaign on water consumption, we adopted a conditional Two-Way Fixed Effects Diff-in-Diff (TWFEDID) strategy. To avoid a potential self-selection bias, we defined through a Propensity Score Matching (PSM) a control group which, before the treatment, was not statistically different from the treated one along a variety of observable dimensions.

We first excluded from our sample non-resident households (which do not live in the apartment for which they hold the water service contract and who could rent their apartment for long or short periods) and those who reported missing or non-consistent water consumption data or information. This allowed us to build a balanced dataset composed by around 9,000 households. Next, the 2 Nearest Neighbor matching procedure brought to select a sub-sample of 2,355 households: 866 treated units and 1,489 control units which did not show any pre-treatment statistical difference with respect to a large variety of dimensions.

Treated and control units were matched only with respect to their observable and available characteristics. This implies that the PSM validity relies on the assumption that matching on observable characteristics allows to match unobservable characteristics. We addressed this potential issue in different ways. First, building on the intuition that individuals sharing similar socio-economic conditions often cluster in the same urban areas – and in light of the small dimension of the municipalities where the information campaign was developed (the average resident population is 24,000 inhabitants) – we proxied several individual-level unavailable characteristics with an array of municipal-level socio-economic, demographic and territorial observable variables. The inclusion of these variables allows to control for several households-level unobservable factors which might cause differences in water consumption among the treated and control groups.

Moreover, in our diff-in-diff model we included time and households-level fixed effects which allow to address a potential omitted variables' bias and to control for potential unobserved heterogeneity. The adoption of a fixed effects model rules out that differences among treated and control units in their water consumption could be potentially driven by unobserved heterogeneity that remains fixed within the time period of our trial.

The validity of our empirical strategy aimed at avoiding a self-selection bias is corroborated by a variety of tests. The PSM balance test shows that in the pre-treatment period treated units are not statistically different from their matched control units with respect to all the considered variables. On top of that, a placebo test and a dynamic analysis show that, in the pre-treatment period, treated units are not statistically different from the matched control units in their water consumption's level and variation. This suggests that the selected treated and control groups do not differ with respect to those unobservable characteristics that correlate and can influence their level and trend of water consumption.

**Findings and Contribution** With this work, we aim to address several research questions. The first concerns the information campaign's effectiveness in nudging a water saving behavior. For this purpose, we developed an Intention to Treat (ITT) analysis where matched treated and control units were compared through a difference-in-differences (DiD) design. In particular, we found that the campaign was effective in promoting an average per capita reduction of 25.8 liters (L) (6,8 gallons) per day compared to the control group. The results of the logarithmic scale analysis suggest that, thanks to the informational campaign, water consumption has decreased by around 6%.

We next developed a dynamic analysis to determine whether the impact of the treatment varied with the number of informative feedback emails sent to the treated units. Indeed, we were interested in establishing whether sending repeated messages on a monthly basis fostered water conservation, thus enhancing water savings over time, or rather, whether the opposite occurred, due to a decrease in consumers' attention over time. We also compared the matched treated and control groups in the months following the end of the campaign to verify whether the campaign was effective in inducing a permanent change in behavior, or whether its effectiveness was temporary and confined to the campaign period.

The dynamic analysis shows that the effect of the campaign was not constant over time, as estimated additional water saving increased with the number of emails sent. However, we verified that the water conservation effects were not permanent and expired a few months after the end of the trial. This suggests that the information campaign did not represent a sufficient tool to drive structural behavioral change.

Third, we develop an Instrumental Variable strategy to estimate the Local Average Treatment effect (LATE) confined to the complying users which effectively opened the received emails. We found that the 25.8 L per day average reduction in water consumption

estimated through the ITT analysis is mainly driven by the complying units. In particular, the LATE estimated with the IV method corresponds to an average per capita water saving equal to 54.9 L per day. In logarithmic terms this corresponds to a 12% reduction in water consumption.

Moreover, we questioned whether the impact of the information campaign varied across the types of consumers and depended on the type of feedback they received. To address this question, we first developed a heterogeneous analysis to assess whether changes in water consumption were uniform across class sizes. We then applied a regression discontinuity design (RDD) around the class sizes' cut-offs to verify whether sending different feedback to units with comparable consumption levels differentially affected their water saving at the margin. This analysis shows that the impact of the treatment was heterogeneous across the consumption classes and that different feedback differentially affected consumption choices at the margin. This result suggests that the social comparison component of the information campaign represents a key driver of the treated units' improved water saving performance.

Our research contributes to the existing literature on water conservation nudging in several ways. The first concerns the information notification tool. Previous researches delivered the information mainly through printed letters, postcards or handouts (Ferraro and Price 2013; Landon et al. 2018; Schultz et al. 2019; Torres and Carlsson 2018; Carlsson et al. 2021; Fielding et al. 2013; Miranda et al. 2020), printed leaflets in the form of door hangers (Schultz et al. 2007; Goette et al. 2019), or a combination of printed letters, emails and a website (Dolan and Metcalfe 2015; Brent et al. 2015; Bhanot 2017; Jessoe et al. 2021; Schultz et al. 2016; Daminato et al. 2021). These studies mainly conclude that the intervention effectiveness depends on the type of notification tool used and find that printed copies tend to be more effective than email notifications, which in general are not associated with a significant effect on water conservation. According to various interpretations, the lower success rate associated with online messages could be due to extra effort required to open them. Conversely, in our case, the information was provided exclusively via email. To the best of our knowledge, this represents the first research providing evidence on the effect of an information campaign based solely on email notifications.

A second novel component of our study is that being entirely developed online, it allowed us to monitor the rate of compliance, proxied by the users' email opening rate, and to differentiate treated units according to their compliance status. We document that the 17% of the treated units never opened any email, while the 44% of the treated units opened a maximum two out of five emails. Due to this non-neglectable non-compliance rate, the ITT approach is likely to underestimate the true treatment effect. Therefore, following previous studies in other fields (Angrist et al. 2010; Leventhal & Brooks-Gunn 2003; Brent and Ward 2019), we complemented the ITT analysis with an Instrumental Variable (IV) model that estimates the Local Average Treatment Effect (LATE) of the information campaign, that is the effect of the treatment restricted to the units who effectively comply with their treatment assignment (Angrist and Pischke 2014). This supportive analysis allows to further verify the robustness of the ITT results and to eliminate the endogeneity of the compliance of underestimation of the treatment while, at the same time, addressing the endogeneity of the compliance decision. To the best of our knowledge, this represents a novel contribution of our research to the existing literature on nudging and water conservation behaviour.

Another way our research contributes to the literature concerns the geographical context. Previous water conservation experiments were mainly developed in the US (Ferraro and Price 2013; Brent et al. 2015, 2020; Schultz et al. 2019), with several applications in various other parts of the world, as discussed in Sect. 2. To the best of our knowledge, ours is the first research applied to the Italian case. Our study aims to determine whether the informative feedback tool, previously used in different contexts, remains effective in a new environment, characterized by an emerging water scarcity problem. Indeed, this is increasingly becoming a critical issue in the South-Europe, due to the conjunction of the climate crisis-induced increase in the frequency and intensity of extreme weather events, such as droughts (EEA 2021, IPCC 2022).<sup>4</sup> In Italy, this criticality is exacerbated by the unsustainable behavior of Italian consumers, who register among the highest level of water consumption in Europe.

Compared to the majority of previous trials, we differentiated our campaign with respect to the type of communicated information. Previous studies mainly communicated the total amount of households' aggregated water consumption over a given period (Ferraro and Price 2013; Brent et al. 2015; Schultz et al. 2019). Other experiments communicated the daily average of households' aggregated water consumption (Bhanot 2017; Jessoe et al. 2021), and rarely the litres per capita per day (Goette, et al. 2019). In our intervention, we communicated the *daily* average water consumption (instead of the total water consumption) at a *per capita* level (instead of at households' aggregated levels). The aim of our choice was to provide information in as familiar terms as possible, so that it could be easily quantified and understood by non-skilled users.<sup>5</sup> We have no tools to assess how clear the communicated information was, and whether the adopted unit of measurement was clearer than other options. Nevertheless, the significant water saving promoted by the information campaign and the relevant estimated LATE effect suggest that the adoption of a per capita daily measure favours a clear understanding of the amount of consumed water, as this measure is associated with a remarkable reduction in water consumption.

The remainder of the paper is organized as follows: Section 2 places our study within the literature on water consumption. Section 3 provides details of the information campaign. Section 4 describes the sample construction and descriptive statistics, in particular it discusses the Propensity Score Matching procedure and the related results. Section 5 presents

<sup>&</sup>lt;sup>4</sup> The Northern Italian regions declared a State of Emergency in the summer of 2022 due to the worst drought in the last 70 years. Since winter 2021, Italy recorded exceptionally low rainfall and snowfall levels. The May-June-July period was among the hottest ever recorded with many temperature anomalies, including very high peaks and heat waves. As a result, by early summer, the snow on the Alps was completely exhausted in Piedmont and Lombardy, and the Po northern river, the longest river in Italy, recorded critical levels of low water. According to the Permanent Observatory on Water Use in Po River Hydrographic District, on June 2022, 'the exhaustion of the flows along the entire Po (...) remains critical throughout the river shaft, with all measurement sections in a condition of extreme severity with flow rates below historical lows (...) with a deviation from the average of over 90%'. Due to the low flow rates, the estimated salt wedge, both in high and low tide conditions, reached maximum saline intrusion values at around 23 km from the mouth in June 2022. On the demand side, Italy records very high levels of per capita water consumption. With an average per capita consumption of 236 L per day, Italy is the second European country in terms of withdrawal of drinking water per inhabitant, against the European average of 144 L per day (ISTAT 2021).

<sup>&</sup>lt;sup>5</sup> As mentioned, water bills report the cumulative volume of cubic meters consumed over a quarter by the entire household. Thus, assuming a given water consumption for a hypothetical three-person family, the water bill would report an aggregated consumption of 55.8 cubic meters over a quarter, while our information campaign would report an average daily water consumption of 200 L per person. Both data refer to exactly the same amount of water consumed over a given period, though the latter communication is undoubtedly easier to understand, since the volume, the related unit of measurement and the time period can be more easily associated with our daily experience. Moreover, compared to the water bill, the higher frequency of our information campaign emails can give users more timely feedback on how any changes in their consumption behavior impacts on their water footprints.

the empirical strategy. Section 6 discusses the results in detail and Sect. 7 shows the robustness checks. Section 8 offers a discussion of specific aspects, and Sect. 9 concludes.

## 2 Literature Review

Actions to promote pro-environmental and resource conservation attitudes have been extensively studied in behavioral science literature (see Andor and Fels 2018 for review). A widely agreed finding is that social information campaigns, on top of being relatively cheap to implement (Wang and Chermak 2021), can be more effective than traditional instruments in promoting sustainable daily habits and stimulating consumers to adopt proenvironmental behaviors (Ferraro and Miranda 2013; Ferraro and Price 2013). This result is confirmed by a variety of field studies, whose designs differ with respect to a variety of factors. We review those that are most strictly connected to our research.<sup>6</sup>

**Geographical context** Existing water conservation experiments were largely and mainly developed in the US (i.e. Ferraro and Price 2013; Brent et al. 2015, 2020; Schultz et al. 2019), with various applications in other parts of the world, such as Australia (Sarac et al. 2003; Fielding et al. 2013), Central and South America (Miranda et al. 2020; Torres and Carlsson 2018), South Africa (Smith and Visser, 2013), and Asia (Agarwal et al. 2017; Goette, et al. 2019). Very few researchers investigated the impact of a social information program on water consumption in Europe (Ansink et al. 2021 in the UK and Kažukauskas et al. 2021 in Sweden).

Adopted Notification Tool Households were reached via postcards or mailers (Fielding et al. 2013; Miranda et al. 2020; Brent et al. 2020), handouts (Seyranian et al. 2015), a combination of letters and emails (Brent et al. 2015; Bhanot 2017; Jessoe et al. 2021), a combination of letters and a website (Schultz et al. 2016; Daminato et al. 2021), printed leaflets in the form of door hangers (Goette et al. 2019) or more frequently via printed letters (e.g., Ferraro and Price 2013; Landon et al. 2018; Schultz et al. 2019; Torres and Carlsson 2018; Carlsson et al. 2021). Few studies provide real-time feedback with pre-installed in-home displays (Kažukauskas et al. 2021), or water meters connected shower

<sup>&</sup>lt;sup>6</sup> These vary, among others, depending on: the inclusion of a social comparison component (Allcott and Rogers 2014; Byrne et al. 2018; Bhanot 2017; Brent et al. 2015, 2020; Jessoe et al. 2021); the type of informative feedbacks sent to the consumers, with a distinction between pure descriptive feedbacks or injunctive normative feedbacks (Bonan et al., 2021; Ferraro and Miranda 2013); the inclusion of a (environmental, economic, social etc.) motivational leverage (Schultz et al. 2007; Jaeger and Schultz 2017); the inclusion of monetary or non-monetary rewards with engagement or gamification approaches (e.g. Brent and Ward 2019; Ferraro and Price 2013; Olmstead and Stavins 2009; Torres and Carlsson 2018; Wichman 2014; Wichman et al. 2016). Also, the frequency of the information campaign differs across studies, and it can be monthly or mixed frequency (Brent et al., 2015; Torres and Carlsson 2018; Carlsson et al. 2021), others last longer than one year (Brent et al., 2015; Brent also differs between studies; some studies last one year (Jessoe et al. 2021), others last one year (Hahn et al., 2015; Bhanot 2017; Torres and Carlsson 2018; Carlsson et al. 2011), others last only a week (Schultz et al. 2017; Torres and Carlsson 2018; Carlsson et al. 2021), and others last only a week (Schultz et al. 2016). We offer more details on the Italian institutional setting and comparison with others in the Appendix A2.

heads (Agarwal et al. 2017).<sup>7</sup> Previous studies show that the effectiveness of an information campaign can depend on how it is communicated. Dolan and Metcalfe (2015) report that printed copies of social norms for electricity conservation are more effective than digital copies delivered via email. Brent et al. (2015) use a combination of letters and emails, and find the effect of their campaign to be insignificant for the category receiving the water report via email. Similarly, Schultz et al. (2016) show that web-based delivery is less effective than postal mail. However, none of these studies provide a solid explanation for this result, suggesting that this could depend mainly on the lower success rate associated with online messages, due to the extra effort required in opening them (Schultz et al. 2016). More recently, using a combination of letters and real-time feedback through an online portal, Daminato et al. (2021) show that the use of an online tool drives the main result of their experiment on water consumption.

**Type of Communicated Information** Recently, Wang and Chermak (2021) argued that the size of the water saving can depend on the unit of measurement being used to communicate consumption data, which varies among studies. While some campaigns communicated the *households' aggregated total amount* of water gallons consumed over a given period (Ferraro and Price 2013; Brent et al. 2015; Schultz et al. 2019) or during the main irrigation season (Landon et al. 2018), others communicated the *daily average* of the households' aggregated total amount of gallons consumed in one or two months (Bhanot 2017; Jessoe et al. 2021). Few experiments used an app or home-installed meters where households could observe their real-time consumption (Agarwal et al. 2017; Kažukauskas et al. 2021). Apart from the notable exception of Goette et al. (2019), to the best of our knowledge, no paper has so far provided information on water consumption expressed both *per person* and *per day* (daily average per capita water consumption), and none have used liters instead of gallons. The liter is the unit of measurement used in Italy. In other countries with the same unit of measurement, data were expressed in cubic meters (Torres and Carlsson 2018; Carlsson et al. 2021), which is a less familiar unit of measurement than liters.

## 3 Information Campaign

Treated units received a monthly report (*the water consumption diary*) on their domestic water use over a five-month period, from September 2021 to January 2022. The report was delivered exclusively via email and included informative feedback and a social comparison component. First, households were informed on their monthly average water consumption, which, differing from previous studies, was communicated on a *per capita* basis using the *liters per day* unit of measurement. Second, we communicated the average per capita water consumption level for the entire treated group, and provided some further information

<sup>&</sup>lt;sup>7</sup> Some studies take advantage of the data collected by using more sophisticated technologies to evaluate the effect of real-time feedback on water consumption. Among these, Agarwal et al. (2017) find that, thanks to real-time consumption feedback for showering, water per shower is reduced on average by 2 L (9-10%) compared to the control group. Analogously, Kažukauskas et al. (2021), making use of pre-installed inhome displays providing real-time information, study the effect of an instant and continuous comparison of consumption on water and electricity for a sample of 525 households in Sweden. They find that, on average, families belonging to the treated units having additional information on their in-home displays reduce their daily energy consumption by an average of 0.3 Kwh (-6.7%), while the only improvement in water consumption is observed in the short term but disappears in the long term.

aimed at facilitating the social comparison in terms of water consumption.<sup>8</sup> In particular, we constructed a ranking on a 1-1000 scale and communicated to each household its ranking position and the related consumption class size: whether households were 'low users', 'medium users' or 'high users' (that is, whether they belonged to the first, second and third consumption tertiles).<sup>9</sup>

Concerning the temporal dimension of the field intervention, the amount of water consumed in a given month (e.g. August) is notified by the smart meter at the beginning of the following month (e.g. September). Therefore, with the information campaign, the email sent in a given month t (e.g. September), necessarily refers to the water consumption registered in the previous month t-1 (e.g. August). We then can observe whether the email sent in the month t promotes a reduction of water consumption during the same month t, as notified by the smart meter at the beginning of the following month t+1 (e.g. October). Figure 1 reports an example of the first email of the information campaign that was sent in September and therefore referred to the water consumption level registered during the previous month, August.

## 4 Design of the Field Intervention and Sample Construction

Few requirements had to be met to be in order to be included in the information campaign. First, households had to be equipped with a smart meter remotely collecting data on their water consumption on a monthly basis.<sup>10</sup> At the time of running our trial, the CAP Holding water company replaced old analogue water meters with electronic smart meters for around 13,800 households distributed over 45 municipalities in the metropolitan area of Milan (Italy). This sample was not subject to self-selection. Households could not apply for smart meters and the water company was not replacing the meters following any pre-determined geographical criteria.

<sup>&</sup>lt;sup>8</sup> The spirit of comparing a household with the behaviour of the average and not of the most virtuous is simply to offer a target that is reasonably achievable. Benchmarking against the most efficient group could discourage the less virtuous portions of the distribution. With reference to this aspect, our design fully aligned with the classical design established by Schultz el al. (2007) where households received feedback about how much energy they had consumed in previous weeks and descriptive normative information about the average consumption of other households in their neighbourhood.

<sup>&</sup>lt;sup>9</sup> A general ranking may not capture a fully accurate distribution of water use, because comparisons are made assuming that households are homogeneous while there are age differences in their compositions. Even though household's age composition might affect water consumption, the literature is divided on which age group consumes the most. A stream of studies (Gregory and Leo 2003; Aprile and Fiorillo 2017) have concluded that older household occupants engage in water conservation behaviours, while Beal and Stewart (2014) and Willis et al. (2013) argue that pensioners consume more water than the average because they tend to spend more time at home. Studies have consistently found that affluent households with children and teenagers tend to experience more frequent and more prolonged micro-component events, such as showers (Makki et al. 2013, 2015; Beal and Stewart 2014; Willis et al. 2013). Abu-Bakar et al. (2023) reported that different age groups show differential water consumption patterns essentially only during peak demand periods.

<sup>&</sup>lt;sup>10</sup> The CAP Holding receives consumption information directly from these devices. We had the opportunity to monitor the actual consumption of all subjects equipped with smart meters (information from the consumer to the supplier), while we also had the option to provide informative feedback exclusively to subjects who, at the time of signing the contract, had given consent to receive additional communications from the water company (information from the supplier to the consumer).

Within this sample, we were legally bound to send informative material only to around 1,000 households, which, at the time of signing their water contract (and thus well before the design of our information campaign), gave their consent to receive informative and ancillary communications from the water company.

A proper randomization within the sample of 1,000 eligible households would have represented the first-best option to assess the causal impact of the information campaign on water saving.<sup>11</sup> Unfortunately, in our case a RCT did not represent a viable option. The CAP water company was committed to involving as many households as possible, therefore it firmly required to send the information campaign to all the 1,000 eligible households. This explicit constraint forced us to adopt an alternative approach to assess the impact of the information campaign.<sup>12</sup>

We initially considered the treated group the entire set of eligible households (those equipped with a smart meter and that had provided privacy consent), while non-eligible households (those equipped with a smart meter, that had not provided privacy consent) were included into the control group.<sup>13</sup>

We believe this selection into the treatment is not likely to raise a relevant endogeneity issue. If we had selected the treated units among those who wanted to participate to an information campaign aimed at reducing water consumption, then we would have created a serious self-selection bias. However, in our case, selection into the treatment was based on a general willingness to receive ancillary communications, with this consent being given before the information campaign. For this reason, the adopted selection criterion is likely to be orthogonal to the intervention condition itself. Nevertheless, we are aware that people voluntarily decided whether or not to give their consent to be advertised. Therefore, following previous relevant literature, we addressed potential self-selection issues through a matching procedure.

#### 4.1 Propensity Score Matching

Robust methodologies have been developed to address potential endogeneity issues and to assess causality when randomization cannot be implemented due to some exogenous constraints. Among them, propensity score matching (PSM) represents a widely adopted

<sup>&</sup>lt;sup>11</sup> This would imply to split the sample into two equivalent groups and to send the information campaign to half sample of randomly selected treated units.

<sup>&</sup>lt;sup>12</sup> This intervention adhered completely to legal requirements, ensuring no discrimination or implicit costs were imposed on consumers. The utility provider reached out via email, sharing additional informative details exclusively with users who had expressly consented to receiving supplementary communication from the company. As the intervention imposed no obligations and solely focused on providing descriptive information, ethical concerns related to this case are not applicable. We did not pre-register the intervention and we regret about our short-sightedness on this relevant aspect. Pre-registration and power analysis are increasingly becoming a pivotal aspect of the planning phase of field interventions. Ex-post sensitivity power analysis reveals that in our setting it would be potentially possible the detection – considering conventional significance level (0.05) and power (0.80) – of small effect size, Cohen's d < 0.2.

<sup>&</sup>lt;sup>13</sup> Consumers in the control group did not receive any feedback. They simply received the conventional bills. At the same time, the consumption levels of all the users included in our study (both treated and control group), were measured on a standard basis by means of the very same smart technology. This is common knowledge in the population of consumers. These measurements were routinely utilized for billing proposes. As a direct consequence, consumers not included in the treatment group were in facts subject to the same constant level of monitoring of their actual consumption levels (Ornaghi and Tonin 2021). This characteristic of the setup contributes to substantial mitigate concerns related to scrutiny/Hawthorne effect.



Fig. 1 The water consumption diary

second-best alternative (Rosenbaum and Rubin 1983; Heckman et al. 1997; Dehejia and Wahba 2002).<sup>14</sup> DeShazo et al. (2017) adopt a PSM technique to assess to which extent the presence of high-occupancy vehicle lanes promotes plug-in electric vehicle adoption. Du and Takeuchi (2019) combine a PSM with the DiD approach to assess whether the renewable energy-based clean development mechanism contributes to poverty alleviation. Recently, Clay et al. (2023) use a difference-in-differences propensity score matching approach to assess the causal impact of LEED (Leadership in Energy & Environmental

<sup>&</sup>lt;sup>14</sup> Rosenbaum and Rubin (1983) propose this method, stating that the propensity score refers to the conditional probability P(Xi) that individual *i* enters the treatment group given a set of covariates (*Xi*). The procedure uses a Logit regression model, Probit and other probability models to estimate the propensity score. The idea is to find a control group that is as similar as possible to the treatment group to reduce selection bias and remove confounding bias of observed variables and other observable factors (Rosenbaum and Rubin 1983). The PSM made the covariates of the treatment and control groups balanced and comparable to control the effect of the treatment.

Design) certification on energy consumption. Cole et al. (2021) use a PSM with a DiD to analyze how firms' carbon intensity is affected by the decision to delocalize some productive activities. Marin et al. (2018) combine the PSM with a DiD to analyze how the implementation of the EU ETS affects the regulated firms' performance compared to the unregulated ones. Castelnovo and Florio (2023) employ a DiD combined with a PSM to assess the impact of public procurement on the patenting activity of the procuring firms in the space sectors, while Clò et al. (2022) combine PSM with a DiD to assess the incremental impact of development banks' support to innovation on behalf of the financed firms. Lin et al. (2023) adopt a PSM-DiD approach to assess the impact of firms' export intensity on their environmental performance.

Following this literature, we selected through a PSM a control group which, before the treatment, was not statistically different from the treated one along a variety of observable dimensions. Treated and untreated units were matched on the estimated propensity scores (on the estimated probability of being treated given a set of observable characteristics on treated and control units). We first estimated through a Logit model to what extent the probability of being treated was explained by a plurality of households-level covariates: *number of residents, age* and *gender* of the contract holder, the *consumer type* (whether it is a final consumer or whether it is a self-employed individual with a proper VAT number) and the aggregate households' pre-treatment water consumption levels (*Cday*).<sup>15</sup> Following Marin et al. (2018), we included the households' pre-treatment variation in water consumption ( $\Delta Cday$ ). The inclusion of this variable in the matching procedure is aimed at ensuring that, in the pre-treatment period, treated and control units show a similar variation in their water consumption. This should therefore allow treated and control households to have parallel trends of the outcome variable before the treatment.

Moreover, we obtained from the CAP Holding water company the information on the latest year customers updated their consensus status (*consensus year*, whether to give consent to receive informative and ancillary communication from the water company) which is the key variable adopted to define whether a family was eligible to receive the treatment. Its inclusion in the matching procedure ensures that the treated and control groups do not show any statistically significant difference with respect to this key variable. In this way, we exclude possible endogeneity issues that could emerge in case the propensity to provide consensus had to vary over time.

It shall be recognized that treated and control units can be matched only with respect to their observable and available characteristics, and thus the PSM validity relies on the assumption that matching on observable characteristics allows to match unobservable characteristics such as income, type of occupation, education or environmental attitudes which are likely to influence their water consumption. To address this potential issue, we proxied some relevant households' unobservable features with municipal-level observable variables. Building on the intuition that individuals sharing similar socio-economic conditions often cluster in the same urban areas – and considering the small dimension of the municipalities where the information campaign was developed (the average resident population is 24,000 inhabitants) – we proxied several individual-level non available characteristics (income, occupation, education) with a variety of socio-economic and territorial

<sup>&</sup>lt;sup>15</sup> Unfortunately, we have no further information on the house's characteristics. Our sample is homogeneous with respect to the type of houses. Indeed, smart meters have been installed exclusively in single-family homes, as they are the only type of residences where it is possible to have precise family-level monitoring. This ensures a certain homogeneity among the families under study.

variables referring to the municipality where households live. Among them, we included: *i*) the municipal average residential *housing prices* as registered by the real estate market observatory (OMI) of the Italian tax authority (Agenzia delle Entrate); *ii*) From the same source, we included information on the total *number of taxpayers* and the total level of declared income per municipality (*total income*); *iii*) from the Bank of Italy database, we extracted information on the *number of bank branches* and on total level of *bank deposits* and *bank loans* at a municipal level; *iv*) the municipal-level *soil consumption* intensity, as provided by Italian Institute for Environmental Protection and Research (ISPRA); *v*) the municipal *migration rate*<sup>16</sup> (ISTAT); *vi*)the municipal *old-age index*<sup>17</sup> (ISTAT); *vii*) the municipal degree of urbanization: whether it is an urban or peripherical area, whether it is a high or low-density building area (source: Italian Department for Development and Economic Cohesion).

Considering that our exercise involved a multitude of small municipalities (on average 24,000 inhabitants), we believe that, overall, municipal-level economic, financial, demographic and urban characteristics allow to proxy for relevant, though non observable, households' characteristics. Indeed, people with higher income and better occupation are likely to live in wealthier municipalities. Therefore, the inclusion of municipal-level variables mitigates the risk that significant differences in water consumption among treated and control units persist due to unobserved differences among them. On top of that, we added among the matching variables also the municipal pre-treatment level of *rainfall* (Agri4cast-JRC). Since water consumption can depend on weather consumption among control and treated units due to exogenous climate conditions.

First, we excluded from our matching procedure those who did not reside in the house equipped with a smart meter (and who could rent their apartment for long or short periods, for instance via Airbnb) or that reported missing or non-consistent water consumption data or ancillary information. This allowed us to work with a balanced dataset composed by 8,741 households, with 866 treated units and 7,875 untreated ones.

The probability of being treated was estimated by:

$$P(Treat_i = 1) = \Lambda \left(\beta X_i + \phi Z_m\right) \tag{1}$$

where P denotes the propensity of households *i* to be treated at time *t*, and  $\Lambda(.)$  is the logistic distribution function. *Treat* is a binary indicator variable that takes a value of 1 if household *i* receives the treatment and 0 otherwise. X is a vector of pre-treatment households' characteristics (see Table 3 for a description), while Z is a vector of municipality-level variables.

From the results reported in Table 1, we can observe that the size of the estimated coefficients (and related marginal effects) is quite small.

<sup>&</sup>lt;sup>16</sup> The municipal migration rate is a demographic indicator that measures the flow of people moving to a given municipality. It is defined by the ratio between the net migration of the year (the total number of people who have moved to the municipality) and the total amount of the resident population.

<sup>&</sup>lt;sup>17</sup> The ISTAT old-age index is a demographic indicator used by the Italian National Institute of Statistics (ISTAT) to assess the distribution of the population across age groups and reflect the demographic structure of a given municipality in relation to the presence of elderly people compared to younger ones. The old-age index is calculated by dividing the number of elderly people (above 65 years) by the number of young people (below 14 years), multiplied by 100.

Based on the estimated propensity scores, we matched each treated unit to a maximum of its two nearest neighbor non-treated units (in terms of estimated propensity score). Non-treated units lying out of the common support of the estimated propensity score were excluded from the analysis. This matching procedure brought to restrict our analysis from the initial sample of around 9,000 households to a sub-sample of 2,355 households, with 866 treated units and 1,489 control units.

A first inspection of the density distribution of the propensity scores in both groups, before and after the matching, visually confirms the common support between treatment and comparison groups, and the soundness of the PSM procedure (see Fig. 2).

The PSM balancing test shows that, along several dimensions, the differences between the treated and the untreated units was significant only before the matching procedure. Conversely, the matched treated and untreated units do not show any statistically significant difference with respect to all the considered variables, thus allowing us to reject the null hypothesis (Table 2).

Table 3 presents the description of the household-level variables and the related summary statistics for the matched sample.

Lastly, we focus on the treatment compliance rate. Having sent the consumption diary via email, we could observe how many of the five monthly emails users received were actually opened by the users (click rate). We cannot verify whether the households which opened the email actually read it. Nevertheless, the email opening rate can be confidently adopted as a confident proxy for the treatment compliance rate. Table 4 reports the email opening rate: the 17% of the treated units never opened any email, while the 44% of the treated units opened a maximum two out of five emails (Table 4).

## 5 Empirical Strategy

The first research question we want to address is whether the information campaign has been effective in reducing water consumption compared to the households that were not involved in the campaign. Since all the treated units were included irrespectively on their compliance status, this corresponds to an Intention To Treat (ITT) analysis, which we address through the following Two-Way Fixed Effects Diff-in-Diff (TWFEDD) model:

$$y_{it} = \alpha + \beta DID_{it} + \gamma_i + u_t + \varepsilon_{it}$$
(2)

where  $y_{it}$  indicates the average daily per capita water consumption for the user *i* at the month *t*. While the post-treatment phase covers the entire period of the information campaign (from September 2021 to January 2022), we decided to restrict the pre-treatment period to the months from May 2021 to August 2021 when there were no COVID-19 restrictions in place. The variable  $DID_{it}$  identifies the unit *i* as belonging to the treated group in the post-treatment period, it corresponds to the interaction term  $TREAT_i \times POST_i$ , where  $TREAT_i$  is a dummy variable equal to 1 when the unit *i* belongs to the treated group and 0 otherwise;  $POST_i$  is a dummy variable equal to 1 in the post-treatment period and 0 otherwise. Its parameter  $\beta$  captures average post-treatment variation in water consumption of the treated group compared to the control group. In our diff-in-diff model we include individual and time and fixed effects, respectively  $\gamma_i$  and  $u_i$ , which allow to address a potential omitted variables' bias and to control for potential unobserved heterogeneity. Indeed, the adoption of a fixed effects model rules out that differences among treated and control units in their water consumption could be potentially driven by unobserved heterogeneity

that remains fixed within the time period of our trial. Finally,  $\varepsilon_{it}$  is the error term, which is clustered at an individual level. Equation (2) is estimated with OLS using the standard fixed effects estimator, with robust standard errors (heteroskedasticity) clustered at a household level.

#### 5.1 Dynamic analysis

A second major interest of our research concerns the dynamic effect of the campaign. We are interested in understanding how the impact of the social information campaign varied over time, whether it increased or decreased with the number of emails sent. The latter case would point to the reinforcing contribution of repeated emails in promoting water conservation behavior, while the former case would point to their limited effectiveness, as water savings would decrease at the margin. Conversely, a constant trend would suggest that sending multiple information campaigns does not affect at the margin water conservation behavior. We address this issue by implement the following event study model:

$$y_{it} = \alpha + \sum_{j=1}^{J} \beta_j LAG_j + \sum_{k=1}^{K} B_k LEAD_k + \sum_{j=1}^{J} \theta_j (LAG_j * TREAT_i) + \sum_{k=1}^{K} \theta_k (LEAD_k * TREAT_i) + \gamma_i + \varepsilon_{it}$$
(3)

Lags and leads are binary variables capturing the months preceding and following the first month of the information campaign. In particular,  $LAG_j$  with j = 1, ..., 4 refers to the months from May 2021 to August 2021, and  $LEAD_k$  with K = 1, ..., 5 are the months from September 2021 to January 2022. The inclusion of lags and leads allows us to assess the dynamic trend of the treatment, whether it is increasing or decreasing in time, whether it is stable or volatile, whether it is permanent or temporary. Moreover, this approach allows us to compare water consumption for the treated group and the control group in the months preceding the launch of the social information campaign, and to test the parallel trend assumption which must be satisfied for the DiD to provide unbiased estimates.

We extend this approach by including observations for the three months following the end of the information campaign (from February 2022 until April 2022). This approach allows us to highlight the differences in water consumption among the treated group and the control group after the end of the information campaign, and to assess whether its effect has been temporary and confined to the treatment period, or whether it managed to induce a structural change in the treated group's behavior, promoting a permanent reduction in their water consumption.

#### 5.2 Compliance and LATE Analysis

In the ITT analysis, all the treated subjects are included according to their original treatment assignment, ignoring noncompliance or withdrawal from the treatment (Hollis and Campbell 1999). The ITT approach preserves the balance between the treated and control groups, allowing for an unbiased estimate of the treatment effect. However, in case of substantial non-adherence, a shortcoming of this approach is a potential untrue estimation of the magnitude of the treatment effect, since non-complying units – which are de facto untreated – are analyzed as if they were treated (Angrist 2006; Gupta 2011).

Being our intervention characterized by a non-neglectable rate of non-compliance, we are interested in complementing the ITT with an analysis able to eliminate the potential source of under-estimation.

#### Table 1 Propensity score estimates

	(1)	(2)
	Estimated coefficients	Marginal Effects
Age	-0.024***	-0.002***
6	(0.003)	(0.000)
Cday	0.000***	0.000***
	(0.000)	(0.000)
ΔCday	-0.030	-0.003
	(0.073)	(0.006)
Consensus vear	0.018***	0.002***
	(0.003)	(0.000)
Gender	0.188**	0.016**
	(0.076)	(0.007)
Number of residents	0.077*	0.007*
	(0.040)	(0.003)
Consumer type	-0.097	-0.008
	(0.750)	(0.065)
Bank branches	0.019	0.002
	(0.046)	(0.004)
Bank deposits	-0.000	-0.000
	(0.000)	(0.000)
Bank loans	0.000	0.000
	(0.000)	(0.000)
Degree of urbanization (city)	0.056	0.005
	(0.458)	(0.040)
Degree of urbanization (suburban, rural)	-0.092	-0.008
	(0.106)	(0.009)
Housing prices	-0.000	-0.000
61	(0.000)	(0.000)
Migration rate	0.027	0.002
8	(0.018)	(0.002)
Old-age index	0.000	0.000
	(0.002)	(0.000)
Soil consumption	-0.006	-0.000
	(0.005)	(0.000)
Taxpayers Number	-0.000	-0.000
	(0.000)	(0.000)
Taxpayers total income	0.000	0.000
	(0.000)	(0.000)
Total rainfall	-0.000	-0.000
	(0.001)	(0.000)
Constant	-37.590***	()
	(6.678)	
No. Obs	8.741	8.741
Pseudo-R2	0.0487	-,

Notes: Logit estimator. For the description of the variables see Table 3. Dependent variable: Treat. Standard errors in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1



Fig. 2 Probability of receiving the treatment before and after the matching

Extending the baseline model (Eq. 2) by further distinguishing treated units according to their compliance status could represent a potential way to capture how the treated units which actually complied with the treatment varied their water consumption in the post-treatment period compared to the control group. However, while allowing to disentangle the average change in water consumption of the treated group among compliers and non-compliers, this would bring to biased estimates due to the self-selection nature of the compliance decision.

According to Angrist and Pischke (2014), the instrumental variable (IV) method – where the treatment assignment is used as an instrumental variable for the treatment effective delivery – eliminates the non-compliance selection bias, thus allowing to "*capture the causal effect of treatment on the treated in spite of the nonrandom compliance decisions made by participants in experiments*". This approach gives an unbiased estimate of the local average treatment effect (LATE), which is the impact of the treatment on compliers-only.

To estimate the LATE we adopt an Instrumental Variable (IV) strategy where the original treatment assignment is used as instrument. This method allows to isolate the variation in actual compliance, which is unrelated to the selection bias, allowing for an unbiased estimate of the treatment effect of compliance on the outcome of interest (Imbens and Angrist 1994).

The IV's first stage assesses to which extent the instrument induces the treated units to effectively take up the treatment. This is estimated by regressing the treatment take up variable  $COMPLY_{it}$  (it equals 1 when the treated household *i* effectively opens the email in the month *t* and 0 otherwise) on the treatment assignment  $TREAT_i$  (it equals 1 when a household receives the consumption diary via email) received in post treatment period:

Table 2         Balance test						
Variables	Source	Matched (M)	Mean		t-test	
		Unmatched (U)	Treated	Control		p >  t
			Households-level va	uriables		
Age	CAP Holding water company	U	56.388	64.401	-14.310	0.000
		Μ	56.388	56.074	0.430	0.667
Cday		U	716.280	641.190	3.850	0.000
		М	716.280	691.100	0.880	0.379
$\Delta C day$		U	0.089	0.095	-0.340	0.732
		Μ	0.089	0.090	-0.050	0.959
Consensus year		U	2004.9	1998.8	11.680	0.000
		М	2004.9	2005	-0.160	0.872
Consumer type		U	1.005	1.005	-0.190	0.846
		Μ	1.005	1.007	-0.450	0.654
Gender		U	0.636	0.602	1.940	0.052
		Μ	0.636	0.649	-0.550	0.582
Number of residents		U	2.657	2.517	4.220	0.000
		Μ	2.657	2.677	-0.420	0.674
			Municipal-level var	iables		
Degree of urbanization	ISTAT	U	2.551	2.588	-1.15	0.249
		Μ	2.551	2.527	0.57	0.567
Migration rate		U	3.964	3.711	2.07	0.039
		М	3.964	3.916	0.28	0.776
Old-age index		U	164.12	165.20	-1.27	0.204
		М	164.12	164.04	0.07	0.944
Soil consumption	ISPRA	U	10.515	10.649	-0.39	0.693
		М	10.515	10.926	-0.92	0.356

_
(pen
ntin
ગુ
2
<u>e</u>
ā
- 63

Table 2 (continued)						
Variables	Source	Matched (M)	Mean		t-test	
		Unmatched (U)	Treated	Control	t	p >  t
Total rainfall	JRC	n	887.96	893.75	-2.3	0.021
		М	887.96	888.61	-0.21	0.83
Bank branches	Bank of Italy	U	5.673	5.568	0.86	0.388
		Μ	5.673	5.660	0.08	0.937
Bank deposits		U	4.8E + 05	4.7E+05	0.78	0.435
		М	4.8E + 05	4.8E + 05	-0.03	0.978
Bank loans		U	3.5E + 05	3.4E + 05	1.08	0.281
		Μ	3.5E + 05	3.5E + 05	0.16	0.87
Housing prices	Ministry of Economy and Finance	U	1670	1659.1	1.2	0.232
		М	1670	1672.4	-0.19	0.852
Taxpayers Number		U	17,575	17,648	-0.21	0.834
		М	17,575	17,752	-0.37	0.708
Taxpayers total income		U	4.30E + 08	4.30E + 08	0.01	0.994
		М	4.30E + 08	4.30E + 08	-0.37	0.714

Table 3 Summary statistics	and description of th	ne variables			
Variable	Mean	Std. Dev	Min	Max	Description
Age <sup>(a)</sup>	56.875	15.174	23	86	Age of the contract holder
Cday <sup>(b)</sup>	603.330	563.408	2.150	9280	Households' aggregated average daily consumption
Cday_pc <sup>(b)</sup>	240.327	287.637	0.716	7781.82	Households' per capita average daily consumption
Consumer type	0.003	0.0544	0	1	Dummy = 1 if the unit is a a self-employed individual with a proper VAT number and 0 if it is a final consumer
Consensus year	2004.472	14.138	1974	2021	Last year when the household updated her consensus status
Gender	0.641	0.479	0	1	Dummy $= 1$ if the contract owner is male
Number of residents <sup>(c)</sup>	2.675	0.969	1	6	Number of residents of each single-family home
Treat	0.368	0.482198	0	1	Dummy = 1 if the household received the information campaign
(a) in years; (b) in liters/day	; (c) in units				

varia
the
of
description
р
statistics ar
Summary
$\sim$

S. Clò et al.

Table 4Descriptive statistics onthe users' email opening rate	Number of opened emails	Number of Users	Percentage	cumulative percentage	
	None (Zero compliance)	147	16.97	16.97	
	One	131	15.13	32.1	
	Two	106	12.24	44.34	
	Three	113	13.05	57.39	
	Four	112	12.93	70.32	
	Five (full compliance)	257	29.68	100	
	Total	866	100		

$$COMPLY_{it} = \beta_0 + \beta_1 DID_{it} + \gamma_i + u_t + \varepsilon_{it}$$
(4.1)

The coefficient  $\beta_1$  captures the variation in compliance determined by the instrument. It can be interpreted as a compliance rate, while the residual term captures the variation in compliance related to self-selection. The first stage equation generates the predicted value  $\widehat{COMPLY}_{it}$  that, being uncorrelated with the error term, is unaffected by the potential source of endogeneity. This approach eliminates the source of bias. In order to be a good instrument, we have to reject the null hypothesis that  $TREAT_i$  is a weak instrument ( $\beta_1$ =0).

In the IV second stage we regress the variable of interest on the predicted value  $COMPLY_{it}$  obtained from the first stage regression:

$$y_{it} = \alpha + \omega C \bar{O} M P \bar{L} Y_{it} + \gamma_i + u_t + \varepsilon_{it}$$

$$(4.2)$$

where the coefficient  $\omega$  captures the LATE. This 2SLS approach isolates that variation in the compliance variable that is unaffected by selection bias and, then, relates it to the outcome of interest. This gives an unbiased estimate of the LATE, that is the unbiased causal effect of the treatment only for the compliers, whose participation in the field intervention is determined exclusively by the treatment assignment.

#### 5.3 Heterogeneity analysis

Next, we develop a heterogeneous analysis to assess whether the treatment effect varies across types of consumers according to some observable characteristics. First, we grouped both treated and control units into tertiles ('low', 'medium' and 'high' classes) according to their pre-treatment average level of per capita water consumption. Then, we extend the baseline model (Eq. 2) by further distinguishing treated units according to their consumption class size j and by estimating the following triple DiD:

$$y_{it} = \alpha + \sum_{j=1}^{3} \varphi_j DID_{it} \times TERTILE_{ij} + \gamma_i + u_t + \varepsilon_{it}$$
(5)

The coefficient  $\varphi_j$  of the triple interaction term captures how the treated units belonging to the consumption tertile *j* varied their water consumption in the post-treatment period compared to the control group. We expect the information campaign effectiveness to depend on the pre-treatment water consumption level, as previously found in some studies on water and energy consumption (Ferraro et al. 2011, Ferraro and Price 2013; Allcott and Rogers 2014, Andor et al. 2020). In particular, consistent with a convex water saving costs function, we expect that high consumers will experience the most significative water consumption reduction, since they should have most water saving opportunities at lower marginal costs. Conversely, consumers belonging to the first water consumption tertile should have limited opportunities to further reduce their water consumption. We therefore expect the information campaign to have limited or no effect on their behavior.

Moreover, we inspect the heterogeneous effects by further distinguishing consumers according to their age and family size. We expect the informative campaign to be more effective for younger users and for smaller families. This is because younger users are expected to be more sensitive to environmental issues. We also expect communication and coordination costs to be lower for smaller families than for larger families. If this is true, then the information campaign should spread more effectively in smaller families, and we should observe a higher reduction in water consumption.

#### 5.4 Feedback and RDD analysis

We are interested in analyzing whether, within the treated group, consumers' behavior varies at the margin depending on the type of feedback received. However, due to the endogenous nature of the informative feedback, a direct comparison across high, medium and low users is likely to lead to a biased estimate of the impact of different feedback on water saving behavior. To address this potential endogeneity issue, we develop a regression discontinuity design (RDD) around the consumption classes' cutoffs. We exploit the fact that consumers were classified into three discrete categories with sharp cutoffs. When all the consumers are considered, the average per capita water consumption differs significantly among classes. Nevertheless, the closer we get to the cutoff, the smaller the difference in consumption between the contiguous categories, with the difference in consumption between households just below and above the cutoff being insignificant. Despite their similarity, consumers around the cutoff are categorized differently and receive different feedback depending on the side of the cutoff they belong to.<sup>18</sup> Therefore, we exploit this quasi-random category assignment among users around the cutoff to estimate the effect of different feedback on water saving behavior. The main intuition of the RDD is that being households just below and above the cutoff similar in their consumption behavior, then any variation in their respective water consumption can be attributed to the different feedback they received.

To implement the regression discontinuity approach, we build a stacked panel dataset in the following way. Within each treatment month t = 1, ..., 5 (from September 2021 to January 2022), we first define  $c_t^{min}$  and  $c_t^{max}$  as the minimum and maximum threshold of the medium consumption category.  $c^{min}$  defines the cutoff between the low-medium classes, while  $c^{max}$  defines the cutoff between the medium-high classes. We then calculate the variables  $D_{it}^{min}$  and  $D_{it}^{max}$  as the differences between each household's consumption (in per capita daily liters), and the  $c_t^{min}$  and  $c_t^{max}$  cutoff points. Then, we consider only the treated units whose distance from the cutoff is lower than a given threshold d, which satisfy the conditions  $D_{it}^{min} \leq d$  and  $D_{it}^{max} \leq d$  respectively for the minimum and maximum cutoffs  $c^{min}$  and

<sup>&</sup>lt;sup>18</sup> The difference in consumption between the best and the worst in the low (middle) consumption category is greater than the difference in consumption between the worst in the low (middle) consumption category and the best in the medium (high) consumption category. Nevertheless, in the first case the two subjects fall in the same low (middle) consumption class size and receive the same feedback despite their significant difference in consumption, while in the second case the two subjects are classified in different categories and receive different feedback despite their substantial similarity in consumption.

 $c^{max}$ . We apply this approach recursively, therefore we construct G=5 groups (corresponding to as many panel datasets), one for each month t=1, ..., 5 of the treatment period. We then stack the G panel datasets and run the following regression:

$$y_{igt} = \alpha + \gamma DID\_RDD_{igt} + \gamma_{ig} + u_{gt} + \varepsilon_{igt}$$
(6)

Notice that the same subject can appear below the cutoff in a certain month and above the cutoff in another month. This implies that the unit of observation is the subject *i* within the group *g*. Therefore, *yigt* represents the water consumption of subject *i*, belonging to group *g*, in the month *t*.  $DID\_RDD_{igt}$  is the interaction term  $RDD_{ig} \times POST\_RDD_t$ , where  $RDD_{ig}$  is a dummy which equals 1 if the subject *i*, belonging to group *g*, falls above the cutoff, and 0 if the subject *i* falls below the cutoff, while  $POST\_RDD_t$  equals 0 in the month when units receive their informative feedback and 1 in the following month. By interacting these two dummy variables we can estimate our parameter of interest  $\gamma$  which captures whether, at the margin, the selected treated units change their water consumption behavior differentially, depending on which side of the cutoff they belong to.  $\gamma ig$  are the fixed effects for the subject *i* within the group *g*, while  $u_{gt}$  are time fixed effects referring to the month *t* within the group *g*.

We first run this regression separately for the two cutoffs  $c^{min}$  and  $c^{max}$  which allows us to compare separately the low category with the medium category, and the medium category to high one. Next we estimate Eq. (6) as a single regression for both cutoffs. This allows to formally test whether the cutoffs matter differentially across the two cutoff types.

## 6 Results

All the models have been estimated employing the dependent variable in both levels and logarithmic terms, aiming to discern the influence of the informational campaign on water conservation, measured respectively in liters per day and in percentage terms.

Table 5 reports the result of ITT analysis obtained by estimating Eq. (2). We find that, on average, the social information campaign had a positive and statistically significant effect on water savings. Indeed, after the treatment, treated units reduced on average their per capita water consumption by 25.8 L per day (6.8 gallons/day) with respect to the control group (Column 1, Table 5). When expressing the dependent variable in logarithmic terms, this corresponds to a water saving equal to 5.7% (see Appendix A1, Table 11).

Our results differ from those of previous studies that did not find a significant effect of information campaigns conducted online and are consistent with the recent finding of Daminato et al. (2021).

#### 6.1 Dynamic analysis

Figure 3 displays the results of the dynamic analysis obtained by estimating Eq. (3). Interestingly, we find evidence that the treatment effect is not constant over time, as it increases with the number of reports sent to the treated units. In particular, after the first email was notified, treated households do not show any statistically significant difference in their water consumption compared to the control group. After the second round of the trial, treated units registered an additional per capita water conservation of 20.4 L (5.4 gallons) per day compared to the control group. Nevertheless, the estimated coefficient was significant only at a 10% level. From the second to the fifth round of the intervention, the treated group registered an additional water saving compared to the control group, which increased from 27.2 L (7.2 gallons) per day in November (significant at a 5% level) to 42 L (11.1 gallons) per day in January (significant at a 1% level).

When the dependent variable is expressed in logarithmic terms, the treated and control groups did not display any statistically significant difference in their water behaviour after the first notified email. Conversely, from the second to the fifth (last) round of the campaign, the treated group registered an additional water savings that increased from -6.9% to -8.7%, both significant at a 5% level (see Appendix A1, Figure 6).

This finding suggests that sending repeated messages enhances higher water savings over time and suggests a rejection of the alternative hypothesis that consumers' attention decreases with the number of messages. Moreover, the figure does not highlight any significant difference in consumers' water consumption among the treated group and the control group in the pre-treatment period. This evidence supports the parallel trend assumption that must be satisfied for the DiD to provide unbiased results.

The long-run analysis reveals that the water conservation induced by the information campaign is not permanent. Indeed, the marginal and significant water savings of the treated group decreases the month after the end of the information campaign while it expired the second month following the end of the campaign. Indeed, we find that no statistically significant difference between the treated group and the control group persisted two months after the end of the campaign. This suggests that per se a five-months information campaign did not represent a sufficient tool to drive structural behavioral change.

#### 6.2 Compliance and LATE Analysis

The ITT analysis was complemented with an analysis aimed at capturing the impact of the information campaign on the adherent units which effectively opened the related emails. To address the potential endogeneity issues associated to the self-section nature of the compliance status we run an IV regression to estimate Eqs. 4.1 and 4.2. This allows us to estimate the Local Average Treatment Effect of the information campaign (Table 6). Results of the IV's first stage show that the coefficient of the instrumental variable is positive and highly significant. By capturing the variation in compliance determined by the instrument, it can be interpreted as a 47% compliance rate.

In the second stage we estimate the effect of the compliance's predicted value on water consumption. The estimated coefficient of interest points to a significant local average treatment effect (LATE), which corresponds to an average per capita water saving of 55 L per day for the complying units compared to the control group. When expressing the dependent variable in logarithmic terms, this corresponds to an additional water saving of 12.2% (see Appendix A1, Table 12). The related endogeneity test rejects the null hypothesis that the compliance variable is exogenous, thus pointing to the need for an IV approach. In the weak identification test, the F value is much larger than the critical value, allowing to reject the null hypothesis that our instrument is weak.

To further inspect the mechanisms underlying the local average effect of the information campaign on the complying treated units, we re-estimate Eqs. 4.1 and 4.2 for different time periods. This setting is aimed at capturing the LATE dynamic trend and brings two interesting results. First, the result of the IV first-stage analysis shows that, when the second informative feedback was sent, the rate of compliance declines from 57.6% to 44.5% and then remains quite stable until the end of the information campaign (see Fig. 4, Panel A). Moreover, we find that the size of the water saving registered by the complying units increases with the length of the post-treatment period – respectively from 1 to 5 months after the beginning of the information campaign – and thus with the number of notifications sent via email. In particular, the first email notification of the information campaign does not induce a significant local average treatment effect. However, this becomes evident and significant after sending the second notification. The additional water saving monotonically increases from 29 L per day to 54 L per day when all the five notifications are considered (Fig. 4, Pane B). This corresponds to an additional water saving which increases from -7.6% to -12% over the same period when expressing the dependent variable in log terms.

On one side, the increased water saving over time suggests that sending repeated feedback helps households to maintain this information salient, allowing to avoid a loss of attention due to cognitive limitations or time constraints. On the other, this result might depend to a certain extent on the change in the complying group composition. Indeed, complying units or households which decide to opt-out from the program might differ from the full study group with respect to some dimensions.

According to Angrist and Hull (2023), while the LATE approach allows to address potential selection bias issues, self-selection into adherence can limit the relevance of IV LATE estimates when specific demographic groups are substantially under-represented among compliers. Therefore, following Angrist and Hull (2023), we compare adherent units with the full study group on a variety of dimensions. Results show that compliers have demographic characteristics broadly representative of the study group at large (see Appendix A4, Table 18). This evidence brings us to exclude the possibility that the increasing water conservation that we observe over time is driven by specific characteristics of non-adherent units that decide to opt-out from the information campaign by not opening the emails that were sent on a monthly basis.

#### 6.3 Heterogeneity analysis

Table 7 reports the results of the triple DiD (Eq. 5), where treated units are further distinguished depending, respectively, on their pre-treatment consumption class (low, medium or high), on the number of resident family members (less than three or more than two), and the age at the time of running our study of the water contract subscriber (under 51 or over 50). We find that the ITT impact is heterogeneous across the consumption classes (Table 7): while the reduction of consumption is not statistically significant for the low and medium categories, the high consumption category strongly reduces its water consumption (-76.8 L/20.3 gallons per day per capita, or a -19%), compared to untreated users belonging to the same category (See Appendix A1, Table 13 for the results of the analysis when expressing the dependent variable in log terms).

It is worth stressing that our study was not specifically designed to test for heterogeneity across types of households. Therefore, the heterogeneity results cannot be fully interpreted in terms of causality. Nevertheless, they suggest that the average reduction in water consumption of the treated group is highly driven by the high consumer category, which has the greatest opportunity to save water at lower marginal costs. Conversely, low level consumers, who already adopt sustainable habits, have higher marginal water conservation costs and do not find significant opportunities to further reduce their water consumption. When exploring heterogeneous effects across age and family sizes (Table 7, Column 2 and 3), we did not find any significant difference among the categories of the treated group. Finally, when considering simultaneously all these categories, we find that

	(1)
	dependent variable: Monthly average per capita water con- sumption
	Liters per day
DID	-25.808**
	(10.494)
Monthly Fixed Effects	
May	65.840***
	(4.428)
June	50.298***
	(5.998)
July	23.049***
	(6.304)
Aug	16.647**
-	(6.726)
Sept	-34.120***
-	(6.595)
Oct	-42.679***
	(6.660)
Nov	-43.920***
	(6.834)
Dec	-28.247***
	(7.449)
Constant	244.836***
	(3.772)
Observations	21,195
Number of id	2,355
Time and unit Fixed Effect	YES
R-squared	0.0564

Table 5 ITT effect on water use

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level. Notes: Month-by-year dummy variables included

the average reduction in water consumption observed for the treated group was mainly driven by the "high-users" consumption class size (Table 7, Column 4).

To provide additional information regarding observable characteristics supporting the heterogeneity analysis, we developed some t-tests to compare household-level characteristics across the different groups (see Appendix A3, Table 17). According to the results, for both the control and treatment groups, users belonging to the high-consumption category are on average older than those belonging to the medium and low-consumption categories. Conversely, medium, and high treated households do not show statistically significant difference in their age. With respect to their gender, low and medium users are highly comparable, and the same holds when comparing medium and high users. Conversely, the share of female is 10% lower in the high-user group compared to the low group, with this



Fig. 3 ITT effect on water use: Dynamic trend. Note: Point Estimate with 95% confidence interval

Table 6Instrumental Variableand LATE effect on water use:		(1)
static analysis	First Stage	
	Treatment Assignment	0.469***
		(0.000)
	Second Stage	Liters per day
	COMPLY	-54.940***
		(11.355)
	Constant	216.104***
		(4.262)
	Observations	21,195
	Time and Individual Fixed Effect	YES
	R-squared	0.0542
	Weak identification test: Cragg-Donald Wald F statistic	6857.775
	Endogeneity test	12.169
	Chi-sq(1) P-value	(0.000)

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level.Notes: Month-by-year dummy variables included

difference being statistically significant. Finally, concerning the number of residents, high users on average have 0.3 residents less than medium and low users.

While being in some cases significant, the differences among consumption classes tend to be quite small. Moreover, it should be highlighted that they represent fixed effects that



Fig. 4 Instrumental Variable and LATE effect on water use: Dynamic Analysis. Note: Point Estimate with 95% confidence interval

we control for in our empirical estimation models and that are included in the matching procedure to increase the comparability among treated and control units. Therefore, these differences are not likely to majorly affect the robustness of our results.

## 6.4 Feedback analysis and RDD

In this section we estimate Eq. (6). Results show that consumers react differently to different feedback (Fig. 5). Indeed, units receiving a 'low user' feedback tend to

	(1)	(2)	(3)	(4)
	Monthly average	water consumption		
VARIABLES	Liters per day			
DID	1.580	-32.456	-15.331	5.158
	(2.834)	(22.956)	(11.350)	(13.850)
DID x Medium	-5.777			-3.275
	(7.709)			(8.211)
DID x High	-76.797***			-72.702***
	(27.863)			(28.184)
DID x Residents $> 2$		11.221		-3.071
		(24.977)		(23.832)
DID x Over 50			-18.057	-5.684
			(19.132)	(16.490)
Constant	279.633***	279.633***	280.336***	280.332***
	(2.636)	(2.834)	(2.839)	(2.643)
Observations	21,195	21,195	21,195	21,195
FE	YES	YES	YES	YES
R-squared	0.102	0.0396	0.0416	0.104

Table 7 ITT effect on water use: Heterogeneity Analysis

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level. Month-by-year dummy variables included

slightly increase their consumption compared to similar units which receive a 'medium user' feedback. An opposite result is found when we compare medium and high consumers around the cutoff: those receiving a 'high user' feedback significantly reduce their consumption compared to similar users who are in the medium category.

A similar finding emerges when the "low-medium" and the "medium-high" cutoffs are considered jointly (Table 8). Running a single regression for both cutoffs allows us to formally test whether the cutoffs matter differentially across the two cutoff types. Also, in this case we find that, around the low-medium cutoff, consumers below the cutoff receiving a "low-users" notification tend to marginally increase their water consumption with respect to medium users which, in spite of having comparable level of water consumption, are located above the cutoff. Similarly, around medium-high cutoff, only those consumers above the cutoff receiving a "high-users" notification tend to marginally reduce their water consumption, while medium users which are located below the cutoff do not register any significant change in their water consumption. This latter result changes when the dependent variable is expressed in log terms (see Appendix A1, Table 14). In this case, we find that both types of consumers reduce in percentage terms their water consumption independently on the side of the cutoff they are located, and thus independently on the type of notification they received. However, those located above the cutoff which received a "high-user" notification reduced their water consumption more than those located below the cutoff.

## 7 Robustness Checks

In this section we present and discuss a battery of analyses developed to further test the validity of the assumptions that must be satisfied in order to support the adopted empirical strategy. Moreover, we want to test whether our findings are robust to alternative empirical strategies.

**Placebo Test** Running a RCT ensures that the units that are randomly assigned to the treatment are not statistically different from the untreated ones. However, being committed to involve into the informative campaign all the eligible households, we could not develop a pure RCT. Given this exogenous institutional constraint, in order to assess the impact of the information campaign we developed a Two-Way Fixed Effects Diff-in-Diff, where the treated and untreated units were selected through a PSM. The related balance tests confirm that, before the treatment, the selected control group did not present any statistically significant difference with respect to the treated one along a variety of dimensions. Moreover, the DID dynamic analysis confirms that, in the months preceding the treatment, the average water consumption of the treated group was not statistically different from that of the control group.

Furthermore, as an additional validation of our results, we perform a placebo test by hypothetically assuming another date for the delivery of the treatment (see Table 9). More precisely, we split the real pre-treatment period into two sub-periods: a pre-treatment period from May 2021 to June 2021 and a false post-treatment period from July 2021 to August 2021. We developed a second placebo test where we take as the pre-treatment period November 2020 to April 2021, and May 2021 to August 2021 as the treatment months. The results of the placebo tests show that, in the false post-treatment period, there are no statistically significant differences in water consumption between the treated and the control group (see Appendix A1, Table 15, for results in log terms). This suggests that treated and control groups did not differ neither with respect to those unobservable characteristics that correlate and can influence their level and trend of water consumption. Moreover, these results provide further evidence supporting the parallel trend assumption that must hold for the staggered difference-in-differences design to provide unbiased estimated.

**Other robustness checks** Hereby, we show that our main findings are confirmed when we introduce some changes to our empirical strategy which bring to a change in the sample composition and size. First, descriptive statistics show that per capita monthly water consumption ranges from a minimum of 0.7 L per day to a maximum of 7,781 L per day. Therefore, we test whether our results were driven from outlier values, namely those households who registered in at least one month a very low level of water consumption (plausibly because they were on vacation); and those who registered too high and anomalous level of water consumption. After trimming the distribution of the per capita water consumption at the 2-nd and 98-th percentile, the average level of per capita daily consumption declines from 240.3 to 211.3 L per day, and now varies from a minimum of 40 L per day to a maximum of 866.6 L per day. As shown in Table 10 (Panel A), when we exclude these outliers from our sample, the results confirm that the treatment has a significant impact on water consumption, though the size of the estimated water saving is now lower, and amounts to 13 L per day, or -4.8% (Appendix A1, Table 16 reports the results when expressing the dependent variable in logarithmic terms).

As a further robustness check, we adopted a stricter matching criterion. In particular, we selected the single closest eligible control unit to be paired with each treated unit (1



Fig. 5 ITT differential impact effect on water use of different feedbacks: RDD. Note: Point Estimate with 95% confidence interval

Nearest neighbor matching). This causes a reduction of the sub-sample from 2,355 to 1,428 households (half of them being treated). Results reported in Table 10 and Table 16 (Panel B) show that our findings are largely confirmed when we adopt an exact 1:1 matching which ensures a higher pre-treatment comparability among the control and treated groups.

We further show that the main findings of the ITT analysis do not depend on the chosen pre-treatment period. Our main findings remain robust when the Eq. (2) is estimated considering a longer pre-treatment time span, from November 2020 to August 2021. Nevertheless, the size of the treatment effect is now smaller and equals to a per capita water saving of 17 L/day compared to the control group, corresponding to a reduction of around 8.8% (see Table 10 and Table 16, Panel C).

We finally test the robustness of our results by running our baseline regression when the entire balanced pre-matching sample is used as control group and by adding household fixed effects and time fixed effects at a household's level. Again, we find consistent results (Table 10 and Table 16, Panel D).

### 8 Discussion

In this section, we address a few aspects that merit specific attention. First, we discuss the information campaign's main drivers of induced water saving. Second, we propose a series of exploratory considerations related to the potential water-saving measures that households could implement in response to the intervention. Third, adopting a comparative

Table 8	ITT effect of feedbacks
on wate	r use: RDD

	(1)	(2)	(3)	(4)
	Monthly aver	rage water con	sumption	
	Liters per day			
Low	6.132	9.807***	12.530***	14.746***
	(3.726)	(2.702)	(2.198)	(1.922)
medium (low)	4.202**	2.370	2.187*	0.762
	(1.970)	(1.503)	(1.320)	(1.108)
medium (high)	3.277	0.953	0.852	0.106
	(4.032)	(2.206)	(1.647)	(1.218)
High	-2.802**	-5.023***	-5.976***	-5.933***
	(1.337)	(1.007)	(0.858)	(0.759)
Constant	168.744***	177.527***	179.311***	183.719***
	(7.568)	(4.606)	(3.532)	(2.845)
Observations	1,076	2,022	2,996	4,038
R-squared	0.074	0.088	0.101	0.108
Cutoff Range	<25	< 50	<75	<100

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses- clustered at household level. Notes: Month-by-year dummy variables included

	(1)	(2)
	Monthly average wa	ater consumption
VARIABLES	Liters per day	
DID	5.436	1.259
	(10.261)	(4.861)
Constant	244.971***	270.443***
	(5.998)	(4.669)
Observations	9,395	20,741
R-squared	0.030	0.024
False Pre-treatment period	2021/05-2021/06	2020/11-2021/04
False Post-treatment period	2021/07-2021/08	2021/05-2021/08

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level. Notes: Month-by-year dummy variables included

approach, we discuss the generalizability of our findings both at national (Italy) and international level.

**Information campaign drivers of water saving** Following previous research in this field, our information campaign, and the message notified via email to the treated users, contained a plurality of information: among others, the monthly average of per capita daily water consumption; the monthly average level of water consumption of the treated group; the household ranking position and the corresponding consumption class size. Our study

# **Table 9**ITT effect on water use:Placebo test

Table 10         ITT effect on water           use: robustness checks		Monthly average	e water consumption
		Panel A	Panel B
		No Outliers	PSM 1:1 Matching
	DID	-13.180***	-36.679***
		(4.563)	(11.811)
	Constant	213.850***	242.265***
		(1.295)	(4.130)
	Observations	18,216	12,852
		Panel C	Panel D
		Longer pre- treatment period	Pre-Matching Balanced sample with Households-level time fixed effects
	DID	-19.785**	-30.470***
		(8.051)	(8.326)
	Constant	210.773***	236.544***
		(3.823)	(0.504)
	Observations	34,744	77,651

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level. Month-by-year dummy variables included

was not designed with the clear intent of assessing the differential impact of each piece of information. Therefore, one limit of our research is that we cannot identify which was precise channel driving the water saving promoted by the information campaign.

Nevertheless, this RDD analysis – where treated units on one side of the cutoff are compared with other treated units that, despite recording similar levels of water consumption, are placed on the other side of the cutoff and therefore receive a different notification of their class size – provides us some interesting insights on the potential driving channel of the treated units' additional water savings. Indeed, the results of the RDD analysis suggest that additional water saving cannot be directly imputed to the provision of an information of which the user was not aware of before the launching of the intervention. Indeed, if the lack of awareness on water consumption was the main cause behind unexploited water savings, then the provision of this information should have promoted additional water saving independently on the side of the cutoff where the consumer was placed.

Conversely, the results of the RDD analysis show that consumers with comparable level of water consumption reacted differently to the information campaign depending on the side of the cutoff they were place. This result suggests that the recorded reduction in water consumption was mainly driven by the social and comparative component of the information campaign rather than by the notification of the amount of notified water consumption per se. In fact, those who received a low consumption notification, and therefore were better labelled than their counterparts who are immediately above the cutoff, increased their consumption at the margin. On the contrary, those who receive a high consumer notification, and therefore performed worse than their counterparts located below the cutoff, reduced their consumption more significantly. This evidence highlights the relevance of the social comparison component of the information campaign in driving a behavioural change. Another open issue concerns the reason why the effect of the campaign vanished a few months after its ending. Again, we cannot unambiguously identify the underlying motivation. Nevertheless, the plurality of our results can provide some useful insights to interpret this evidence. One potential explanation could be that, due to the adopted measurement unit (per capita daily water consumption) people did not really understand the message embedded in the campaign. Indeed, this information differed from the one reported in the water billing (households aggregate water consumption per quarter). We do not think this explanation to be exhaustive. Indeed, if the provided information was not clear enough (due to the chosen measurement unit), we should not have observed any additional water saving during the period of the campaign. Conversely, we document that, when it was in place, the campaign promoted a significant reduction in water consumption.

Another potential explanation is that the campaign, in spite of being effective in bridging a cognitive gap on the users' water footprint, was too short to drive a structural behavioral change. This potential explanation is consistent with previous evidence on the attention bias and on the memory retrieval bias. Concerning the former, due to cognitive limitations, time constraints, or simply because certain information is more salient or easier to process than others, people can focus only on a limited set of information or stimuli, while ignoring or undervaluing other relevant information, and therefore make sub-optimal choices (Cowan 1988, 1995; Styles 2005). The latter bias refers to the tendency of individuals to recall and retrieve information from memory in a biased or selective manner (de Fockert JM 2005; Furely and Wood 2016). Moreover, the ending of the social comparison with the end of the campaign can partly explain the fact that the savings promoted by the campaign were not persistent over time. Being the social norm component so relevant in driving our result, the not persistent water saving can depend by the short timing of the intervention, which did not favour the emergence of a social norm of water conservation.

**Sources of the positive impact of the intervention** Even though our setting does not offer any opportunity to investigate at a micro-level how water is actually used and consumed by household's members, we can exploratory discuss which actions can drive a reduction in water consumption, but we cannot determine which of them were specifically triggered by the intervention. These actions can be clustered into two broad categories, depending on whether water consumption declines as an effect of a change in behavior or a technological change.<sup>19</sup> According to the study by Ansink et al (2021) which focused the on impact of audits on water conservation, specifically distinguishing between the information and technological components, we know that the adoption of devices reducing water pressure are particularly effective in the long run while information component of the water audit has a large initial impact, but this gradually fades. Considering the relatively mild and time-limited nature of our intervention, coupled with the absence of explicit emphasis on technological components, we are inclined to suggest that the water savings triggered by our intervention primarily stem from a heightened adoption of micro water-saving behaviors. This includes, but is not limited to, practices such as shorter showers and running full loads

<sup>&</sup>lt;sup>19</sup> According to descriptive estimates, a faucet left open releases on average between 10 and 13 L per minute, while applying an aerated flow restrictor can reduce the flow to 5 L per minute. Handwashing dishes typically requires an average of 40 L, but a modern dishwasher consumes approximately 8 L of water per load. A modern washing machine consumes an average of 40/50 L of water per load, whereas older models can consume up to 120 L. A traditional toilet flush contains from 10 to 16 L. Proper use of the dual flush button allows saving 8–10 L per flush. These average estimates can vary significantly depending on the type of technology which is in place.

in the dishwasher, as well as a proper use of toilet flush. We would tend to exclude that an intervention of such nature induced households to buy new and more efficient appliances or significantly upgrading aspects of the domestic hydraulic system.

External validity of evidence generated through this study As argued by Andor et al. (2020), generalizing the results of a filed study of this sort to other contexts can be challenging. Along a similar line, Vivalt (2020) concludes that the extent to which experimental results can be generalized remains an open question. Like other water consumption studies in the existing literature, our intervention has different characteristics regarding the context, study design, empirical setup, sample size, intervention studied, and treatment. Nevertheless, our findings align with those from other geographic and institutional contexts. Existing literature on household water conservation indicates a reduction in water consumption ranging between 4 and 9% due to nudge programs. Notable references include studies by Carlsson et al. (2021), Schultz et al. (2019), Torres and Carlsson (2018), Miranda et al. (2020), and Jessoe et al. (2021). In essence, the results of our analyses are comparable to those obtained from trials conducted in various geographical and institutional contexts. The consistency of results across studies conducted in different locations suggests a certain external validity of the nudging approach. This is promising as it indicates the potential effectiveness and generalizability of nudge strategies in promoting water conservation in different contexts. As part of our study, we recognize that although our findings are in line with other studies, they cannot be generalized to the entire European population. However, they can be extended to European cities with similar characteristics. The full generalizability of the results is limited by several considerations. Compared to other Italian areas, the Milan metropolitan area is richer and more dynamic, making it more similar to some European cities than to other Italian cities. On the other hand, the levels of water consumption per capita are high and not comparable to those of other European cities, but closer to those recorded in other areas of Italy. Indeed, the average water consumption in the metropolitan city of Milan aligns with that of other Italian cities, where water stress is also increasing. In detail, in the Italian metropolitan cities, where over 30% of the Italian population resides, the daily water supply provided for authorized uses reach an overall average of 236 L per inhabitant,<sup>20</sup> one of the highest water usages per person in Europe.<sup>21</sup> These differences may be influenced by factors such as climate, lifestyle, water management practices, and individual awareness of water conservation. Italy's context, characterized by a relatively higher water consumption, offers an intriguing scenario for investigating the potential impacts of informational campaigns on domestic water usage. This underscores the necessity of tailored initiatives to promote water conservation. On top of that, we believe that the region

<sup>&</sup>lt;sup>20</sup> In northern regions, such as Lombardy and Veneto, where industrialization and urbanization are more pronounced, households may tend to have higher water consumption due to a more developed economy and lifestyle. In contrast, southern regions may experience lower average household water consumption, influenced by a warmer climate and potentially different living habits. In particular, there is notable heterogeneity in water consumption at the Italian level. According to the data on water supplied for authorized uses per capita in liters per day for the year 2020 (Istat, 2021), daily per capita water consumption figures for different regions in Italy highlight notable variations, with the North West, which includes Milan, registering a relatively high consumption of 253 L per inhabitant per day. Moving to the North East, the figure remains substantial at 212 L, displaying a noticeable but slightly lower consumption pattern. The Central region records a per capita consumption of 199 L, while the South and Islands exhibit figures of 203 and 186 L, respectively. This data highlights the regional disparities in water availability and consumption across Italy.

gal showed a moderate range with figures of 117 and 111 L, respectively.

where we developed our analysis shares with other areas an increasing issue of water scarcity linked to the climate crisis' induced droughts, as exemplified by Catalonia's recent declaration of a water emergency. This mirrors Italy's past experiences and echoes the challenges witnessed across the continent. Catalonia, notably, is bracing itself for a drought emergency. Our results are likely to be less applicable in countries where water consumption is already low, but they may hold relevance in areas that, due to escalating droughts stemming from the climate crisis, must take actions to reduce consumption. However, we anticipate that external validity may diminish in different settings, such as smaller Italian cities or European cities where per capita water consumption is considerably lower than the Italian average.

## 9 Conclusions

With this research we analyzed the effects of a water informative campaign run from September 2021 to January 2022 on a sample of around 1,000 households, equipped with smart meters and located in the metropolitan area of Milan. The informative campaign was designed to bring some contributions to the existing literature. First, it is the only campaign developed in Italy, a country characterized by an increasing water scarcity issue, due to both the intensifying of climate-related droughts and to the unsustainable behavior of the Italian citizens, who, on average, consume 236 L (62.3 gallons) of water per day, one of the highest levels in Europe. Second, compared to previous trials, we communicated the *daily* average water consumption (instead of the total water consumption) at a *per capita* level (instead of the households' aggregated levels), thus providing information in a format as familiar as possible, so that it could be easily quantified and assimilated by nonskilled users. The third contribution of our research related to the information notification tool. Unlike previous studies, we decided to provide the information on water consumption exclusively via email, which allowed us to monitor the email click (open) rate and the related opt-out rate. Thanks to this, we could adopt the ITT analysis as our main analysis and complement it with a LATE estimation through an IV approach.

The main result of our research suggests that the information campaign was effective in promoting an average reduction in per capita water consumption equal to 25.8 L/day (6.8 gallons/day), corresponding to a 6% water saving. Considering the intervention involved around 1,000 households with an average of 2.6 inhabitants, this roughly corresponds to around 2 million liters (528 thousand gallons) saved per month and more than 10 million liters (2.6 million gallons) over the period of the intervention. It is important to note that this number refers to the average (ITT) effect of the information campaign regardless of its compliance rate. Moreover, we show that this effect is mainly determined by users who have been effectively treated. Indeed, when focusing on compliers, we estimate a 55 L per day water saving (LATE). We further compared adherent and non-adherent units with the full study group across a variety of dimensions (see Appendix A4). Since various demographic groups are well-represented among LATE compliers, we can conclude that the IV estimates of the information campaign's effect are likely to predict these effects beyond the group that produced them. This evidence brings us to exclude the possibility that the increasing water conservation that we observe over time is driven by specific characteristics of non-adherent units that decide to opt-out from the information campaign by not opening the emails that were sent on a monthly basis.

Existing literature on domestic water conservation studies indicates a reduction in water consumption ranging between 4 and 9% due to nudge programs (Torres and Carlsson 2018; Schultz et al. 2019; Miranda et al. 2020; Carlsson et al. 2021, and Jessoe et al. 2021). In our study, in response to the intervention, per capita water consumption decreased by around 6%. Compared to the existing literature, we employed an information campaign based exclusively via email, communicated a different type of information, operated in a different geographical context, and additionally, had the opportunity to observe the compliance status, allowing us to estimate the local average treatment effect.

We believe that our results expand on the previous literature which found that website campaigns were little or no effective in promoting water conservation habits and reveals that paperless online campaigns can be an effective and cheap instrument in promoting water saving behavior.

Our information campaign, and the message notified via email to the treated users, contained a plurality of information (among others the per capita level of water consumption, corresponding consumption class size, the average water consumption). Therefore, one limit of our research is that we cannot identify which was precise channel driving the water saving promoted by the information campaign. Nevertheless, the results of the RDD analysis suggest that water saving was not promoted just by the pure information on the level of water consumption. Indeed, consumers with comparable levels of water consumption reacted differently to the information campaign depending on the side of the cutoff they were place and on the related feedback they received. This result suggests that the change in behavior was largely influenced by the type of feedback received.

Some further considerations can be developed by estimating the economic savings associated to the lower water consumption promoted by the information campaign. In Italy, the water sector is regulated by the national Authority ARERA. According to the Italian regulation, the water tariff is divided into two parts. The fixed quota includes the water service tariffs for aqueduct, sewerage and purification, i.e. the management costs of the water network. The related rate is established by the territorial Agency and can vary from province to province. In Lombardy, this varies not only for the type of user but also for the different typologies of private users (final consumers or self-employed individuals with a proper VAT number). The variable quota varies depending on the number of resident people composing a household, the actual per capita water consumption, and the application of any exemption or concession applied to particularly vulnerable class of consumers. The CAP Holding water company estimated that the average cost per cubic meter of water for a 4-members family belonging to the category"residents' water domestic use" is € 1.37 (or € 0.034 per liter per capita), for a total expense of approximately €450 per year.<sup>22</sup> With a simplified simulation, we can estimate that, if the 25 L per day of water saving promoted by the our intervention was permanent, this would have corresponded to a per capita annual monetary saving equal to  $\notin 3.12$ . For a 4-members family, the annual water tariff would decline by  $\notin 12.5$ . This would correspond to a 3% decline in the annual water bill. Considering that the field intervention lasted for five months only and that its effect vanished few months after its end, and considering the lower number of residents belonging to the treated group, the real monetary saving associated to the information campaign was actually modest. This suggests that the water saving per se, and the related (quite modest) economic saving do

<sup>&</sup>lt;sup>22</sup> https://acquadelrubinetto.gruppocap.it/salute/quanto-costa-metro-cubo-acqua/

not seem to represent the main channel of the change in water behaviour promoted by the intervention.

Finally, the dynamic analysis highlighted some interesting insights. Indeed, while we found that the amount of water saving increased with the number of emails over the long term the water conservation impact of the campaign was not permanent, and it expired a few months after the end of the trial. This suggests that information campaign did not represent a sufficient tool to drive a structural behavioral change and puts into doubt whether the observed water saving was effectively induced by a correction of the users' cognitive biases on their consumption levels, or whether it was induced by the feeling of being observed. Again, we are unable to precisely identify the main reason why water saving expired a few months after the information campaign. We think that the reason cannot be attributed to the notification tool or to the type of provided information, in particular to the possibility that people did really understand the message embedded in the campaign. Indeed, if it was the case, we should not have observed any water saving during the period of the campaign. Conversely, we document that, when it was in place, the campaign promoted a significant reduction in water consumption. On top of this consideration, the results of the RDD analysis suggest that the main reason behind the vanishing effect of the campaign few months after its end cannot be attributed to the lack of environmental awareness or by the lack of updated information on water consumption. Indeed, the RDD analysis shows that information on the level of water consumption was not per se the main driven of the observed water saving. Indeed, people with similar levels of water consumption reacted differently to the information campaign depending on the feedback they received. Therefore, the social comparison component of the campaign, and in particular the relative positioning of each user with respect to the average consumer, played a significant role in influencing water consumption behavior. In a similar vein, the fact that the social comparison ended with the end of the campaign can partly explain the fact that the savings promoted by the campaign were not persistent over time. Moreover, being the social norm component so relevant in driving our result, the not persistent water saving can depend by the short timing of the intervention, which did not favour the emergence of a social norm of water conservation.

We believe these findings to have some relevant policy implications. Given the current and growing water emergency that is affecting advanced and non-advanced countries, our results show that the implementation of information campaigns, through more advanced electronic tools, would allow the policymakers to achieve the objectives of reducing water consumption in line with the United Nations' SDGs. Furthermore, public policies are needed to encourage water companies to adopt information campaigns. Government guarantees, soft loans and tax incentives are some standard examples. However, our results also suggest the importance of complementing these information campaigns with other measures aimed at inducing a structural change towards more sustainable behavior.

## Appendix A1

#### Alternative Results of the Empirical Strategy: Dependent Variable in Log Terms

This Appendix reports the main results of our analysis when the main dependent variable (Monthly average per capita water consumption) is expressed in logarithmic terms.

Tables 11, 12, 13, 14, 15 and 16

Consumption Feedback and Water Saving: A Field Intervention...

Table 11ITT effect on wateruse – Dependent Variable in logterms

	(1)
DID	-0.057***
	(0.021)
Time Fixed Effects	
May	0.187***
	(0.008)
June	0.089***
	(0.011)
July	-0.057***
	(0.014)
Aug	0.032**
	(0.013)
Sept	-0.142***
-	(0.013)
Oct	-0.189***
	(0.013)
Nov	-0.199***
	(0.013)
Dec	-0.141***
	(0.014)
Constant	5.272***
	(0.007)
Observations	21,195
Number of id	2,355
Time and Individual Fixed Effect	YES
R-squared	0.111

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level. Notes: Month-by-year dummy variables included



Fig. 6 ITT effect on water use: Dynamic trend – Dependent Variable in log terms. Note: Point Estimate with 95% confidence interval

<b>Table 12</b> Instrumental Variableand LATE effect on water use:		(1)
static analysis – Dependent	- First Stage	
variable in log ternis	Treatment Assignment	0.469***
		(0.000)
	Second Stage	Log Value
	COMPLY	-0.122***
		(0.025)
	Constant	5.130***
		(0.009)
	Observations	21,195
	Time and Individual Fixed Effect	YES
	R-squared	0.110
	Weak identification test: Cragg-Donald Wald F statistic	6857.775
	Endogeneity test	12.169
	Chi-sq(1) P-value	(0.000)

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses—clustered at household level

Notes: Month-by-year dummy variables included

Table 13         ITT effect on water           use:         Heterogeneity Analysis –		(1) Monthly av	(2)	(3)	(4)
Dependent Variable in log terms	VARIABLES	Log Value		consumption	
	DID	0.023 (0.028)	-0.066* (0.036)	-0.060* (0.033)	0.020 (0.040)
	DID x Medium	-0.051 (0.038)			-0.052 (0.038)
	DID x High	-0.191*** (0.048)			-0.193*** (0.048)
	DID x Residents $> 2$		0.012 (0.045)		-0.027 (0.039)
	DID x Over 50			-0.000 (0.043)	0.031 (0.037)
	Constant	5.327*** (0.005)	5.327*** (0.006)	5.332*** (0.006)	5.332*** (0.005)
	Observations	21,195	21,195	21,195	21,195
	FE R-squared	0.142	1ES 0.0644	1ES 0.0710	0.145

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses – clustered at household level. Notes: Month-by-year dummy variables included

Table 14ITT effect of feedbackson water use:RDD – DependentVariable in log terms

	(1)	(2)	(3)	(4)
	Monthly ave	erage water co	nsumption	
	Log Value			
Low	0.012	0.030	0.055***	0.076***
	(0.025)	(0.019)	(0.014)	(0.012)
medium (low)	0.004	-0.007	-0.008	-0.015**
	(0.011)	(0.009)	(0.007)	(0.006)
medium (high)	-0.014	-0.016***	-0.013***	-0.015***
	(0.009)	(0.006)	(0.005)	(0.004)
High	-0.021***	-0.029***	-0.032***	-0.031***
	(0.005)	(0.004)	(0.004)	(0.003)
Constant	5.144***	5.171***	5.169***	5.178***
	(0.020)	(0.015)	(0.012)	(0.011)
Observations	1,076	2,022	2,996	4,038
R-squared	0.100	0.126	0.138	0.147
Cutoff Range	<25	< 50	<75	<100

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses- clustered at household level. Notes: Month-by-year dummy variables included

	(1)	(2)
	Monthly average wa	ater consumption
VARIABLES	Log Value	
DID	0.023	0.016
	(0.024)	(0.015)
Constant	5.274***	5.237***
	(0.015)	(0.016)
Observations	9,395	20,741
R-squared	0.081	0.030
False Pre-treatment period	2021/05-2021/06	2020/11-2021/04
False Post-treatment period	2021/07-2021/08	2021/05-2021/08

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses-clustered at household level. Notes: Month-by-year dummy variables included

Table 16         ITT effect on water           use: robustness checks –		Monthly averag	e water consumption
Dependent variable in log terms		Panel A	Panel B
		No Outliers	PSM 1:1 Matching
	DID	-0.048***	-0.076***
		(0.018)	(0.025)
	Constant	5.256***	5.271***
		(0.005)	(0.009)
	Observations	18,216	12,852
		Panel C	Panel D
		Longer pre- treatment period	Pre-Matching Balanced sample with Households-level time fixed effects
	DID	-0.033**	-0.076***
		(0.017)	(0.017)
	Constant	5.127***	5.222***
		(0.008)	(0.001)
	Observations	34,744	77,651

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. Robust standard errors in parentheses-clustered at household level. Notes: Month-by-year dummy variables included

Table 15 ITT effect on water use: Placebo test - Dependent Variable in log terms

## Figure 6 Appendix A2 Italian Institutional Setting of the Water Sector

The water sector, akin to other sectors of public interest, has witnessed a process of restructuring and reforms across Europe, encompassing changes in market organization, market competition, vertical integration of companies, and ownership structure. Despite a unified European regulatory framework, the implementation intensity of these reforms varies among countries. Unlike many sectors, the water industry remains a local natural monopoly, limiting competition among companies within the same territorial area. Market opening has taken the form of competition for the market, with water services previously managed in-house now outsourced through tendering procedures. Notably, public entities have transformed into private entities, diversifying the ownership structure of water companies. This includes traditional publicly-owned, mixed enterprise models, and private enterprises.

For instance, the CAP Holding water company, a private legal entity, is entirely owned by local public entities, specifically the municipalities of the metropolitan area of Milan. As a public entity, the CAP Holding water company pursues social objectives, including sustainability.

Cap Holding had limited possibility to promote this goal via price mechanism. Given its natural monopoly characteristics, the water sector is a highly regulated sector. The Regulatory Authority for Energy, Networks, and the Environment (ARERA), an independent administrative authority in Italy, plays a pivotal role in regulating the water sector. ARERA oversees tariff matters, identifying efficient investment and operating costs, determining criteria for tariff determination, and approving proposed tariffs. It also ensures accounting transparency through proper cost and revenue disaggregation (unbundling). ARERA's functions extend to quality oversight, setting minimum levels and objectives for integrated water service quality, and enforcing automatic compensation obligations for violations.

For water companies, the price channel is not utilized to incentivize water savings due to regulation. The price covers costs but excludes capital remuneration from the tariff components. ARERA, in tariff regulation, defines a fair, transparent, and non-discriminatory tariff system. It determines cost components, periodically revises tariff determination methods, and approves tariffs.

Comparatively, major European countries exhibit diverse approaches, ownership structures, and regulatory frameworks in their water sectors. France employs a mixed system regulated by the Ministry of Ecological Transition and Water Agencies, while Germany's predominantly publicly owned and decentralized sector contrasts with Italy's local monopoly. The UK's privatization under Ofwat regulation, Spain's decentralized water sector, and the Netherlands' emphasis on decentralized water management through water boards further underscore the variations in water management approaches across Europe. These differences result from unique historical developments, policy choices, and regulatory structures in each country.

## **Appendix A3**

This Appendix reports the t-test on household-level variables compared across the different groups: low, medium and high users. The analysis is reported within both the control and treatment groups.

Table 17

Table 17 T-test	t on household	d-level variables	within the differ	ent group	s analyzed							
CONTROL GR	OUP											
	LOW vs M	<b>(EDIUM USERS</b>	74		LOW vs HI	<b>GH USERS</b>			<b>MEDIUM vs HI</b>	<b>IGH USERS</b>		
	Low Users	Medium Users	difference (a)—(b)	SE	Low Users	High Users	difference (a)—(b)	SE	Medium Users	High Users	difference (a)—(b)	SE
Age	55.195	57.754	-2.559**	0.980	55.195	58.530	-3.335***	0.988	57.754	58.530	-0.776	0.931
Gender	0.648	0.665	-0.017	0.030	0.648	0.619	0.029	0.031	0.665	0.619	0.046	0.030
Number of residents	2.622	2.768	-0.146*	0.061	2.622	2.597	0.025	0.062	2.768	2.597	0.171**	0.054
Number of users	497	496			497	496			496	496		
Total number of users	993				993				992			
TREATED GI	ROUP											
	LOW vs M	<b>JEDIUM USERS</b>	74		LOW vs HI	<b>GH USERS</b>			<b>MEDIUM vs H</b>	<b>IGH USERS</b>		
	Low Users	Medium Users	difference (a)—(b)	SE	Low Users	High Users	difference (a)—(b)	SE	Medium Users	High Users	difference (a)—(b)	SE
Age	53.488	56.986	-3.498**	1.219	53.488	58.698	$-5.210^{***}$	1.248	56.986	58.698	-1.712	1.239
Gender	0.588	0.637	-0.048	0.041	0.588	0.684	-0.096*	0.040	0.637	0.684	-0.047	0.039
Number of residents	2.716	2.775	-0.059	0.084	2.716	2.479	0.237**	0.084	2.775	2.479	0.296***	0.079
Number of users	289	289			289	288			289	288		
Total number of users	578				577				577			

\*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1

🙆 Springer

## **Appendix A4**

Table 18 compares compliers with the full study group under different definitions of compliance. High compliers are those treated units which opened at least 3 out of the 5 emails of the information campaign. Always compliers are those treated units which opened all the 5 emails of the information campaign, while we define as opt-out treated units those households which opened the first email received in September 2021 but that did not open the second email received the following months. High compliers do not show any statistically significant difference with the full study group with respect to all the considered dimensions. Always compliers and opt-out units show a difference with respect to the full group, respectively, with respect to the gender and number of residents. In particular, always compliers show a slightly higher percentage of males (7 percentage points), while opt-out units are composed by a slightly lower number of residents (-0.17). Although there are some differences within these sub-samples, both demographic groups are well-represented among compliers.

To further analyze the determinants of the opt-out decision, we estimated a probit model where the probability of opening the email received in the month t is explained by a series of observable households' characteristics, time and unit fixed effects, and by the type of feedback received in the previous month t-1. Results reported in Table 19 show that, apart from the number of residents, none of the estimated coefficients of the explanatory variables are highly statistically significant. In light of this evidence, we can assert that the opt-out decision does not depend on the type of informative feedback they received nor on the units' characteristics. We only find that the probability of opening the email increases with the number of residents.

Tables 18 and 19

study group	
and full s	
compliers a	
variables among	
T-test on household-level v	
Table 18	

	High Compliers vs	Full Study group			Always Compliers	vs Full Study group			Opt-out treated uni	ts vs Full Stu	dy group	
	Full study group	High Compliers	t	p >  t	Full study group	Always Compliers	t	p >  t	Full study group	Opt-out treated units	t	p >  t
Cday	625.66	646.04	0.82	0.41	629.19	635.15	0.18	0.85	638.52	649.5	0.24	0.81
Cday_pc	251.08	253.57	0.2	0.84	250.71	258.76	0.49	0.62	253.69	284.67	1.42	0.16
Residents	2.65	2.71	1.18	0.24	2.66	2.63	-0.46	0.64	2.69	2.52	-2.02	0.04
Gender	0.63	0.67	1.63	0.1	0.63	0.69	1.82	0.07	0.64	0.64	0.05	0.96
Age	56.65	57.75	1.42	0.16	56.75	57.87	1.12	0.26	56.28	56.85	0.43	0.66

<b>Table 19</b> Probit estimates: Thedeterminants of openness	Variables	Estimated coefficients
	Low class (informative feedback)	0.069
		(0.102)
	Medium class (informative feedback)	-0.039
		(0.089)
	Residents	0.115**
		(0.058)
	Gender	0.199
		(0.125)
	Age	0.005
		(0.004)
	Constant	-0.424
		(0.683)
	Observations	3,448

\*\* p < 0.05. Robust standard errors in parentheses

Acknowledgements We thank the CAP Holding water company for having shared their water consumption data and for their willingness to develop this field intervention. We are grateful to the Editor and three anonymous reviewers for their remarks and suggestions that helped us improve the paper substantially. We thank Andrea Geraci and the participants to the EAERE workshop on behavioural ecological economics, to the IAERE 2022, and to the A.I.S.Re 2023 annual conferences for their useful comments. We are the only responsible for any potential mistake.

Funding Open access funding provided by Università degli Studi di Firenze within the CRUI-CARE Agreement.

## Declarations

**Conflict of Interest** The authors certify that they have NO affiliations with or involvement in any organization or entity with any financial interest (such as honoraria; educational grants; participation in speakers' bureaus; membership, employment, consultancies, stock ownership, or other equity interest; and expert testimony or patent-licensing arrangements), or non-financial interest (such as personal or professional relationships, affiliations, knowledge or beliefs) in the subject matter or materials discussed in this manuscript.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

## References

Abu-Bakar HH, Williams L, Hallett SH (2023) Contextualising household water consumption patterns in England: A socio-economic and socio-demographic narrative. Cleaner and Responsible Consumption 8:100104

- Agarwal S, Goette L, Sing TF, Staake T, Tiefenbeck V (2017) The Role of Goals and Real-Time Feedback in Resource Conservation: Evidence from a Large-Scale Field Experiment. National University of Singapore, Singapore
- Allcott H, Rogers T (2014) The short-run and long-run effects of behavioral interventions: Experimental evidence from energy conservation. Am Econ Rev 104:3003–3037. https://doi.org/10.1257/aer.104.10. 3003
- Andor MA, Fels KM (2018) Behavioral economics and energy conservation A systematic review of nonprice interventions and their causal effects. Ecol Econ 148:178–210. https://doi.org/10.1016/j.ecole con.2018.01.018
- Andor MA, Gerster A, Peters J, Schmidt CM (2020) Social norms and energy conservation beyond the US. J Environ Econ Manag 103:102351
- Angrist JD (2006) Instrumental variables methods in experimental criminological research: what, why and how. J Exp Criminol 2:23
- Angrist JD, Pischke J-S (2014) Mastering' metrics: The path from cause to effect. Princeton University Press
- Angrist JD, Dynarski SM, Kane TJ, Pathak PA, Walters CR (2010) Inputs and impacts in charter schools: KIPP Lynn. Am Econ Rev 100:239–243
- Angrist JD, Hull P (2023) Instrumental variables methods reconcile intention-to-screen effects across pragmatic cancer screening trials. Proceedings of the national academy of sciences 120(51):e2311556120
- Ansink E, Ornaghi C, Tonin M (2021) Technology vs information to promote conservation: Evidence from water audits. Tinbergen Institute Discussion Papers 21–014/VIII
- Aprile MC, Fiorillo D (2017) Water conservation behavior and environmental concerns: Evidence from a representative sample of Italian individuals. J Clean Prod 159:119–129
- Beal CD, Stewart RA (2014) Identifying residential water end uses underpinning peak day and peak hour demand. J Water Resour Plan Manag 140(7):04014008
- Benartzi S, Beshears J, Milkman KL, Sunstein CR, Thaler RH, Shankar M, Tucker-Ray W, Congdon WJ, Galing S (2017) Should governments invest more in nudging? Psychol Sci 28(8):1041–1055
- Bhanot SP (2017) Rank and response: A field experiment on peer information and water use behavior. J Econ Psychol 62:155–172
- Bonan J, Cattaneo C, d'Adda G, Tavoni M (2021) Can social information programs be more effective? The role of environmental identity for energy conservation. J Environ Econ Manag 108:102467
- Brent DA, Ward MB (2019) Price perceptions in water demand. J Environ Econ Manag 98:102266. https://doi.org/10.1016/j.jeem.2019.102266
- Brent DA, Cook JH, Olsen S (2015) Social comparisons, household water use, and participation in utility conservation programs: Evidence from three randomized trials. J Assoc Environ Resour Econ 2:597–627
- Brent DA, Lott C, Taylor M, Cook J, Rollins K, Stoddard S (2020) What causes heterogeneous responses to social comparison messages for water conservation? Environ Resource Econ 77:503–537. https:// doi.org/10.1007/s10640-020-00506-0
- Byrne DP, La Nauze A, Martin LA (2018) Tell me something I don't already know: Informedness and the impact of information programs. Rev Econ Stat 100(3):510–527
- Carlsson F, Jaime M, Villegas C (2021) Behavioral spillover effects from a social information campaign. J Environ Econ Manag 109:102325
- Castelnovo S. Clò, Florio M (2023) A quasi-experimental design to assess the innovative impact of public procurement: an application to the italian space industry. Technovation 121:102683
- Clay K, Severnini E, Sun X (2023) Does LEED certification save energy? Evidence from retrofitted federal buildings. J Environ Econ Manage 121:102866
- Clò S, Frigerio M, Vandone D (2022) Financial support to innovation: The role of European development financial institutions. Res Policy 51(10):104566
- Cole MA, Elliott RJ, Okubo T, Zhang L (2021) Importing, outsourcing and pollution offshoring. Energy Economics 103:105562
- Cowan N (1988) Evolving conceptions of memory storage, selective attention, and their mutual constraints within the human information processing system. Psychol Bull 104(2):163–191
- Cowan N (1995) Attention and Memory: An Integrated Framework. Oxford Univ. Press, New York
- Daminato C, Diaz-Farina E, Filippini M, Padrón-Fumero N (2021) The impact of smart meters on residential water consumption: Evidence from a natural experiment in the Canary Islands. Resource and Energy Economics 64:101221
- de Fockert JM (2005) Keeping priorities: the role of working memory and selective attention in cognitive aging. Sci Aging Knowl Environ 44:pe34

- Dehejia R, Wahba S (2002) Propensity score-matching methods for nonexperimental causal studies. Rev Econ Stat 841:151–161
- DeShazo JR, Sheldon TL, Carson RT (2017) Designing policy incentives for cleaner technologies: Lessons from California's plug-in electric vehicle rebate program. J Environ Econ Manage 84:18–43
- Dolan P, Metcalfe RD (2015) Neighbors, knowledge, and nuggets: two natural field experiments on the role of incentives on energy conservation (CEP Discussion Paper No. 1222). Retrieved from the London School of Economics Centre for Economic Performance website: http://eprints.lse.ac.uk/ 51563/1/dp1222.pdf
- Du Y, Takeuchi K (2019) Can climate mitigation help the poor? Measuring impacts of the CDM in rural China. J Environ Econ Manag 95:178–197
- Ebi KL, Bowen K (2016) Extreme events as sources of health vulnerability: Drought as an example. Weather Clim Extremes 11:95–102
- EEA (2021) Water resources across Europe Confronting water stress: An updated assessment, EEA Report, 12/2021, European Environment Agency
- European Commission (2022) Consequences of Climate Change. Retrieved June 30, 2022, from: https:// ec.europa.eu/clima/climate-change/consequences-climate-change\_en
- European Union (2020) Green City Accord Explanatory Note on Monitoring and Reporting & Set of Mandatory Indicators. Retrieved January 15, 2024, from: https://environment.ec.europa.eu/docum ent/download/43622fbe-3052-41dc-aedc-44277391c5a3\_en?filename=EN\_GCA\_mandatory% 20indicators\_FINAL\_02052022.pdf
- Ferraro PJ, Miranda JJ (2013) Heterogeneous treatment effects and mechanisms in information-based environmental policies: Evidence from a large-scale field experiment. Resour Energy Econ 35:356– 379. https://doi.org/10.1016/j.reseneeco.2013.04.001
- Ferraro PJ, Price MK (2013) Using nonpecuniary strategies to influence behavior: Evidence from a large-scale field experiment. Rev Econ Stat 95:64–73
- Ferraro PJ, Miranda JJ, Price MK (2011) The persistence of treatment effects with norm-based policy instruments: Evidence from a randomized environmental policy experiment. Am Econ Rev 101:318–322
- Fielding KS, Spinks A, Russell S, McCrea R, Stewart R, Gardner J (2013) An experimental test of voluntary strategies to promote urban water demand management. J Environ Manag 114:343–351
- Franzke CLE (2021) Towards the development of economic damage functions for weather and climate extremes. Ecol Econ 189:107172
- Furley P, Wood G (2016) Working memory, attentional control, and expertise in sports: A review of current literature and directions for future research. J Appl Res Memory Cogn
- Goette L, Leong C, Qian N (2019) Motivating household water conservation: A field experiment in Singapore. PLoS ONE 14(3):e0211891
- Gregory GD, Di Leo M (2003) Repeated behavior and environmental psychology: the role of personal involvement and habit formation in explaining water consumption. J Appl Social Psychol 33.6:1261–1296
- Gupta SK (2011) Intention-to-treat concept: A review. Perspect Clin Res 2:109–112. https://doi.org/10. 4103/2229-3485.83221
- Hollis S, Campbell F (1999) What is meant by intention to treat analysis? Survey of published randomised controlled trials. The BMJ 319:670–674
- Imbens GW, Angrist JD (1994) Identification and estimation of local average treatment effects. Econometrica 62:467–475
- IPCC (2022) Climate change 2022: impacts, adaptation and vulnerability
- ISTAT (2021) Le Statistiche dell'Istat sull'acqua, anni 2018–2020. Down from: https://www.istat.it/it/ archivio/255596. Accessed 30 May 2024
- ISTAT (2022) Le Statistiche dell'Istat sull'acqua, anni 2019–2021. Down from: https://www.istat.it/it/ archivio/268242. Accessed 30 May 2024
- Jaeger CM, Schultz PW (2017) Coupling social norms and commitments: testing the under detected nature of social influence. J Environ Psychol 51:199–208
- Jessoe K, Lade GE, Loge F, Spang E (2021) Residential water conservation during drought: Experimental Evidence from Three Behavioural Interventions. J Environ Econ Manag 110:102519. https://doi. org/10.1016/j.jeem.2021.102519
- Kažukauskas A, Broberg T, Jaraitė J (2021) Social comparisons in real time: A field experiment of residential electricity and water use. Scand J of Econ 123:558–592. https://doi.org/10.1111/sjoe.12422
- Landon AC, Woodward RT, Kyle GT, Kaiser RA (2018) Evaluating the efficacy of an information-based residential outdoor water conservation program. J Clean Prod 195:56–65
- Leventhal T, Brooks-Gunn J (2003) Moving to opportunity: An experimental study of neighborhood effects on mental health. Am J Public Health 93:1576–1582

- Lin X, He LY, Bateman IJ (2023) The more the merrier? Evidence from firm-level exports and environmental performance in China. Environ Resour Econ 84(1):125–172
- Makki AA, Stewart RA, Panuwatwanich K, Beal C (2013) Revealing the determinants of shower water end use consumption: enabling better targeted urban water conservation strategies. J Clean Prod 60:129–146
- Marin G, Marino M, Pellegrin C (2018) The impact of the European emission trading scheme on multiple measures of economic performance. Environ Resour Econ 71:551–582
- Miranda JJ, Datta S, Zoratto L (2020) Saving water with a nudge (or two): Evidence from Costa Rica on the effectiveness and limits of low-cost behavioral interventions on water use. World Bank Econ Rev 34(2):444–463. https://doi.org/10.1093/wber/lhy025
- Olmstead SM, Stavins RN (2009) Comparing price and nonprice approaches to urban water conservation. Water Resour Res 45(4):W04301
- Ornaghi C, Tonin M (2021) The effects of the universal metering programme on water consumption, welfare and equity. Oxf Econ Pap 73(1):399–422
- Rosenbaum PR, Rubin DB (1983) The central role of the propensity score in observational studies for causal effects. Biometrika 70:41–55
- Sarac K, Day D, White S (2003) What are we saving anyway? The results of three water demand management programs in NSW. Australia Water Supply 3(3):215–222. https://doi.org/10.2166/ws.2003.0029
- Schultz PW, Nolan JM, Cialdini RB, Goldstein NJ, Griskevicius V (2007) The constructive, destructive, and reconstructive power of social norms. Psychol Sci 18(5):429
- Schultz PW, Messina A, Tronu G, Limas EF, Gupta R, Estrada M (2016) Feedback and the moderating role of personal norms: a field experiment to reduce residential water consumption. Environ Behav 48:686–710
- Schultz W, Javey S, Sorokina A (2019) Social comparison as a tool to promote residential water conservation. Front Water. https://doi.org/10.3389/frwa.2019.00002
- Seyranian V, Sinatra GM, Polikoff MS (2015) Comparing communication strategies for reducing residential water consumption. J Environ Psychol 41:81–90
- Styles EA (2005) (2005) Attention, perception, and memory. Psychology Press, Hove, UK, An integrated introduction
- Sunstein CR (2018) 'Better off, as judged by themselves': a comment on evaluating nudges. Int Rev Econ 65:1–8. https://doi.org/10.1007/s12232-017-0280-9
- Thaler RH, Sunstein CR (2008) Nudge: Improving Decisions About Health, Wealth, and Happiness. Yale University Press, New Haven, CT
- Torres MMJ, Carlsson F (2018) Direct and spillover effects of a social information campaign on residential water-savings. J Environ Econ Manag 92:222–243
- Vivalt E (2020) How much can we generalize from impact evaluations? J Eur Econ Assoc 18(6):3045–3089 Wang J, Chermak JM (2021) Is less always more? Conservation, efficiency and water education programs.
- Ecol Econ 184:106994. https://doi.org/10.1016/j.ecolecon.2021.106994
- Wichman CJ (2014) Perceived price in residential water demand: Evidence from a natural experiment. J Econ Behav Org 107:308–323
- Wichman C, Taylor L, von Haefen R (2016) Conservation policies: Who respond to price and who responds to prescription? J Environ Econ Manag 79:114
- Willis RM, Stewart RA, Giurco DP, Talebpour MR, Mousavinejad A (2013) End use water consumption in households: impact of socio-demographic factors and efficient devices. J Clean Prod 60:107–115
- World Bank (2015) World development report 2015: Mind, Society, and Behavior. Washington, DC

**Publisher's Note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.